

# Working Paper Series 284

## On the Welfare Costs of Business-Cycle Fluctuations and Economic-Growth Variation in the 20<sup>th</sup> Century

Osmani Teixeira de Carvalho Guillén, João Victor Issler and Afonso Arinos de Mello Franco-Neto July, 2012

					50 00.038.166/0001-0
Working Paper Series	Brasília	n. 284	Jul.	2012	p. 1-44

ISSN 1518-3548 CGC 00.038.166/0001-05

## Working Paper Series

Edited by Research Department (Depep) - E-mail: workingpaper@bcb.gov.br

Editor: Benjamin Miranda Tabak – E-mail: benjamin.tabak@bcb.gov.br Editorial Assistant: Jane Sofia Moita – E-mail: jane.sofia@bcb.gov.br Head of Research Department: Adriana Soares Sales – E-mail: adriana.sales@bcb.gov.br

The Banco Central do Brasil Working Papers are all evaluated in double blind referee process.

Reproduction is permitted only if source is stated as follows: Working Paper n. 284.

Authorized by Carlos Hamilton Vasconcelos Araújo, Deputy Governor for Economic Policy.

#### **General Control of Publications**

Banco Central do Brasil Secre/Comun/Cogiv SBS – Quadra 3 – Bloco B – Edifício-Sede – 1° andar Caixa Postal 8.670 70074-900 Brasília – DF – Brazil Phones: +55 (61) 3414-3710 and 3414-3565 Fax: +55 (61) 3414-3626 E-mail: editor@bcb.gov.br

The views expressed in this work are those of the authors and do not necessarily reflect those of the Banco Central or its members.

Although these Working Papers often represent preliminary work, citation of source is required when used or reproduced.

As opiniões expressas neste trabalho são exclusivamente do(s) autor(es) e não refletem, necessariamente, a visão do Banco Central do Brasil.

Ainda que este artigo represente trabalho preliminar, é requerida a citação da fonte, mesmo quando reproduzido parcialmente.

#### **Consumer Complaints and Public Enquiries Center**

Banco Central do Brasil Secre/Surel/Diate SBS – Quadra 3 – Bloco B – Edifício-Sede – 2° subsolo 70074-900 Brasília – DF – Brazil Fax: +55 (61) 3414-2553 Internet: <a href="http://www.bcb.gov.br/?english">http://www.bcb.gov.br/?english</a>

## On the Welfare Costs of Business-Cycle Fluctuations and Economic-Growth Variation in the 20th Century<sup>\*</sup>

Osmani Teixeira de Carvalho Guillén<sup>†</sup> João Victor Issler<sup>‡</sup> Afonso Arinos de Mello Franco-Neto<sup>§</sup>

#### Abstract

Lucas(1987) has shown a surprising result in business-cycle research: the welfare cost of business cycles are very small. Our paper has several original contributions. First, in computing welfare costs, we propose a novel setup that separates the effects of uncertainty stemming from business-cycle fluctuations and economic-growth variation. Second, we extend the sample from which to compute the moments of consumption: the whole of the literature chose primarily to work with post-WWII data. For this period, actual consumption is already a result of counter-cyclical policies, and is potentially smoother than what it otherwise have been in their absence. So, we employ also pre-WWII data. Third, we take an econometric approach and compute explicitly the asymptotic standard deviation of welfare costs using the Delta Method.

Estimates of welfare costs show major differences for the pre-WWII and the post-WWII era. They can reach up to 15 times for reasonable parameter values –  $\beta = 0.985$ , and  $\phi = 5$ . For example, in the pre-WWII period (1901-1941), welfare cost estimates are 0.31% of consumption if we consider only permanent shocks and 0.61% of consumption if we consider only transitory shocks. In comparison, the post-WWII era is much quieter: welfare costs of economic growth are 0.11% and welfare costs of business cycles are 0.037% – the latter being very close to the estimate in Lucas (0.040%). Estimates of marginal welfare costs are roughly twice the size of the total welfare costs. For the pre-WWII era, marginal welfare costs of economic-growth and businesscycle fluctuations are respectively 0.63% and 1.17% of per-capita consumption.

<sup>\*</sup>We gratefully acknowledge the comments of Caio Almeida, Larry Christiano, Carlos E. da Costa, Wouter den Haan, Robert F. Engle, Pedro C. Ferreira, Antonio Fiorencio, Clive Granger, Rodolfo Manuelli, Samuel Pessoa and Octavio Tourinho on this or on earlier versions of this paper. All errors are ours. We thank CNPq-Brazil, FAPERJ and INCT for financial support.

<sup>&</sup>lt;sup>†</sup>osmani.guillen@bcb.gov.br, Banco Central do Brasil and Ibmec-RJ, Brazil

<sup>&</sup>lt;sup>‡</sup>jissler@bcb.gov.br, Graduate School of Economics - EPGE Getulio Vargas Foundation.

<sup>&</sup>lt;sup>§</sup>amfranco@bcb.gov.br, Graduate School of Economics - EPGE Getulio Vargas Foundation.

The same figures for the post-WWII era are, respectively, 0.21% and 0.07% of per-capita consumption.

JEL Codes: E32; C32; C53

Keywords: Business cycles fluctuations, economic growth variation, welfare costs, structural time-series model.

#### 1. Introduction

From the perspective of a representative consumer, who dislikes systematic risk, it makes sense for macroeconomic policy to try to reduce the variability of pervasive shocks affecting consumption. The best known welfare-cost approach to this issue was put forth by Lucas (1987, 3), who calculates the amount of extra consumption a rational consumer would require in order to be indifferent between an infinite sequence of consumption under uncertainty (aggregate consumption) and a consumption sequence with the same deterministic growth and no cyclical variability. Here, business-cycle shocks are the only source of variation for aggregate consumption. Thus, Lucas' measure is known as the *welfare cost of business cycles*. For 1983 figures, using a reasonable parametric utility function (CES or Power utility function), and post-WWII data, the extra consumption is about \$ 8.50 per person in the U.S., a surprisingly low amount.

Several papers have been written just after Lucas first presented his results. For example, Imrohoroglu (1989) and Atkeson and Phelan (1995) recalculated welfare costs using models with a specific type of market incompleteness. Van Wincoop (1994), Pemberton (1996), Dolmas (1998), and Tallarini (2000) have either changed preferences or relaxed expected utility maximization. In some of them, welfare costs of business cycles reached up to 25% of per-capita consumption. On that matter, Otrok (2001) notes that "it is trivial to make the welfare cost of business cycle as large as one wants by simply choosing an appropriate form for preferences."

Regarding the original setup, as in Zellner's (1992) version of the KISS principle, Lucas *Keeps It Sophisticatedly Simple*: if only transitory shocks hit consumption, the best a *macroeconomist* can hope to achieve in terms of welfare improvement is to eliminate completely its cyclical variation, which is equivalent to eliminating all systematic risk. Of course, the implicit counter-factual exercise being performed is rather extreme, since no one really believes that this trained *macroeconomist* can indeed eliminate all cyclical variation in consumption. Shutting out completely the uncertainty behind shocks to consumption in computing welfare costs forces the counter-factual exercise to be of limited practical importance. Moreover, it dismisses any sources of uncertainty affecting long-term growth. Indeed, Lucas recognizes that the setup could also include permanent shocks, which lead Obstfeld (1994) to compute welfare costs in this context; see also Dolmas, Tallarini, Issler, Franco, and Guillén (2008) and Reis (2009), the latter showing explicitly the importance of properly measuring the persistence in aggregate consumption.

In a very interesting paper, Alvarez and Jermann (2004) generalized the setup in Lucas by proposing a more realistic counter-factual exercise, where the representative consumer is offered a convex combination of consumption and its conditional mean, but not a deterministic sequence *a priori*. Their setup includes the *total* and the *marginal* welfare costs of business cycles. Total welfare costs are computed when, in the counter-factual exercise, all the weight goes to the conditional mean as in Lucas<sup>1</sup>. Marginal costs are obtained when we consider small changes in welfare costs in the neighborhood of observed consumption, which has a more practical appeal.

More recently, the literature has focused on rare disasters – Barro (2009); on the effects of model uncertainty on the welfare cost of business cycles – Barillas, Hansen, and Sargent (2009); on how the stochastic properties of aggregate consumption affects welfare cost estimates – Reis; on the distinction between individual and aggregate consumption risk in computing welfare costs – De Santis (2009); and on the difference between welfare costs based on preference-parameter values that fit or not asset-pricing data – Melino (2010).

In our view, despite the existence of a seemingly mature literature, there are still important issues to be discussed in it. Consider models where aggregate consumption is hit by permanent shocks (shocks affecting economic growth) and transitory shocks (typical business-cycle shocks). The nature and sources of these shocks are completely different and they can arise in the real-business-cycles tradition, e.g.,

<sup>&</sup>lt;sup>1</sup>To be equivalente to the exercise in Lucas, all the weight should go the unconditional mean instead. However, Alvarez and Jermann want to take into account the possibility that consumption is non-stationary. Thus, they focus on the conditional mean, which is still well defined in this case.

King, Plosser and Rebelo (1988), and King et al. (1991), or in new-keynesian tradition, e.g., Galí (1999). As we note in a previous paper (Issler, Franco, and Guillén), the welfare impact of permanent and transitory shocks is completely different: for the former, its conditional variance increases without bound with time, whereas it is bounded for the latter. Hence, separating the effects of these two type of shocks in a sensible way requires thinking deeper about the counter-factual exercise being performed. An easy solution is to lump all uncertainty together, computing the welfare costs of what we have labelled *macroeconomic uncertainty*. However, this approach is clearly limited in scope, given the very different roles that these two types of shocks play and their potentially different sources. Indeed, this dichotomy has been key in macroeconomics and in macro-econometrics since the seminal work of Phelps (1967, 1968, 1970).

Another important issue that deserves further attention is the fact that (almost) all of the previous literature has computed welfare costs for the post-WWII period<sup>2</sup>. Although this is interesting on its own right, it helps little in measuring the welfare benefits of counter-cyclical policies, for the simple reason that they were already in place during this period. Borrowing ideas from the *treatment-effect* literature, post-WWII aggregate consumption reflects already the *treatment* from counter-cyclical policies, thus it cannot serve as a benchmark to compute the welfare benefits associated with them. One candidate to compute the latter is to use pre-WWII consumption data, which lead us to compute here "The Welfare Costs in the 20th Century." We recognize that the match is not perfect, since the pre-war period may include policies (or lack thereof) that hurt welfare. Despite that, it is interesting on its own right: separating the samples in pre-WWII and post-WWII allows to measure by how much welfare costs have changed over time, something that could serve as a guide for current and future macroeconomic policy.

Our paper has three original contributions. First, while the whole literature

 $<sup>^{2}</sup>$ The only exception is Alvarez and Jermann, who also estimated welfare costs including the pre-WWII period (1889-2001 and 1927-2001), although they do not present separate pre- and post-WWII results. In any case, their emphasis is on the post-WWII period (1954-2001).

makes no effort to construct a setup that separates the effects of uncertainty stemming from business-cycle fluctuations and economic-growth variation, we explicitly make an effort to do so. In addition to that, uncertainty is computed in a bivariate model containing consumption and income, which enlarges the conditioning set used by the representative consumer in extracting consumption shocks, something that is not seen in the literature. Here, permanent shocks to consumption arise from the unit-root component in its trend. There are empirical reasons for that, e.g., Hall (1978), Nelson and Plosser (1982), Engle and Granger (1987), King et al. (1991), Issler and Vahid (2001), and Reis. There are also theoretical reasons: in the consumption literature - e.g., Hall (1978) and Flavin (1983) - it is shown that consumption should follow a martingale; in the stochastic discount factor literature - e.g., Alvarez and Jermann (2005), and Hansen and Scheikman (2009) - it is shown that the limit stochastic discount factor must entail permanent shocks. Indeed, as stressed by Alvarez and Jermann, "for many cases where the pricing kernel is a function of consumption, innovations to consumption need to have permanent effects." Thus, we model the trend in consumption as martingale process to accommodate this need. The fluctuations about the trend (the cycle) are modelled as a stationary and ergodic zero-mean process. Trend and cyclical innovations are assumed to be independent, which allows the joint measurement of welfare costs of business cycles and of economic-growth variation.

Second, we depart from Lucas in changing the sample from which to compute the moments of consumption: the whole of the literature chose primarily to work with post-WWII data. However, for this period, actual consumption is already a result of counter-cyclical policies, and is potentially smoother than what it otherwise have been in their absence. It would be desirable to measure the welfare cost of business cycles observed in times with no (or little) counter-cyclical policy. Despite the caveat raised above, that is why we use pre-WWII data.

Third, we take an econometric approach, and compute explicitly the asymptotic standard deviation of welfare costs using the Delta Method. This allows us to compute confidence bands for welfare costs. Indeed, we go back to the idea behind the original exercise done by Lucas, where he notes that: "It is worth re-emphasizing that these calculations rest on assumptions about preferences *only*, and not about any particular mechanism – equilibrium and disequilibrium – assumed to generate business cycles." In other words, we need not specify a full structural model to investigate the welfare costs of business cycles in the presence of trend and cyclical shocks, which is exactly our approach.

The paper is divided as follows. Section 2 provides a theoretical and statistical framework to evaluate the welfare costs of business cycles. Section 3 provides the estimates that are used in calculating them. Section 4 provides the calculations results, and Section 5 concludes. There is also an Appendix providing the econometric background necessary to implement the calculations carried out in the paper.

#### 2. The Problem

Lucas (1987) proposed the following way to evaluate the welfare gains of cycle smoothing (or the welfare costs of business cycles). Suppose that consumption  $(c_t)$  is *log-Normally* distributed about a deterministic trend:

$$c_t = \alpha_0 \left(1 + \alpha_1\right)^t \exp\left(-\frac{1}{2}\sigma_z^2\right) z_t, \qquad (2.1)$$

where  $\ln(z_t) \sim N(0, \sigma_z^2)$  is the stationary and ergodic cyclical component of consumption.

Cycle-free consumption is the sequence  $\{c_t^*\}_{t=0}^{\infty}$ , where  $c_t^* = \mathbb{E}(c_t) = \alpha_0 (1 + \alpha_1)^t \exp\left(-\frac{1}{2}\sigma_z^2\right) \mathbb{E}(z_t) = \alpha_0 (1 + \alpha_1)^t$ . Notice that  $\{c_t^*\}_{t=0}^{\infty}$  is the resulting sequence when we replace the random variable  $c_t$  with its unconditional mean. Hence, for any particular time period,  $c_t$  represents a mean-preserving spread of  $c_t^*$ .

An intuitive way of thinking about  $c_t^*$  is realizing that:

$$c_t^* = \lim_{\sigma_z^2 \to 0} c_t = \lim_{\sigma_z^2 \to 0} \alpha_0 \left( 1 + \alpha_1 \right)^t \exp\left( -\frac{1}{2} \sigma_z^2 \right) z_t = \alpha_0 \left( 1 + \alpha_1 \right)^t.$$

Hence,  $c_t^*$  is a degenerate random variable with all the mass of its distribution at  $\alpha_0 (1 + \alpha_1)^t$ , obviously risk free.

Risk averse consumers prefer  $\{c_t^*\}_{t=0}^{\infty}$  to  $\{c_t\}_{t=0}^{\infty}$ . Then, to evaluate the welfare costs of business cycles, amounts to calculating  $\lambda$ , which solves the following equation<sup>3</sup>:

$$\mathbb{E}\left(\mathbb{E}_{0}\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1+\lambda\right)c_{t}\right)\right) = \sum_{t=0}^{\infty}\beta^{t}u\left(c_{t}^{*}\right),\tag{2.2}$$

where  $\mathbb{E}_t(\cdot) = \mathbb{E}(\cdot | \mathcal{I}_t)$  is the conditional expectation operator of a random variable, using  $\mathcal{I}_t$  as the information set,  $u(\cdot)$  is the utility function of the representative agent who discounts future utility at the rate  $\beta$ . Then, the welfare cost is expressed as the compensation  $\lambda$ , that consumers would require at all dates and states of nature, which makes them indifferent between the uncertain stream  $\{c_t\}_{t=0}^{\infty}$  and the risk-free stream  $\{c_t^*\}_{t=0}^{\infty}$ .

Notice that uncertainty here comes in the form of stochastic business cycles alone, since the trend in consumption is purely deterministic. One important limitation of this setup is that it prevents the existence of permanent shocks to consumption. Of course, at least since Nelson and Plosser (1982), macroeconomists have benefitted from the dichotomy of having econometric models with permanent and transitory shocks, the first being associated with permanent factors influencing economic growth - such as productivity, population, etc., and the second being associated with transient factors - such as monetary policy.

Since Lucas modelled consumption trend as deterministic, eliminating all the cyclical variability in  $\ln(c_t)$  is equivalent to eliminating all its variability. Under difference stationarity for (log) consumption, where the econometric model now entails a permanent-transitory decomposition for shocks, this equivalence is lost, since uncertainty comes both in the trend and the cyclical component of  $\ln(c_t)$ . Moreover,  $\mathbb{E}(c_t)$  is not defined, since the stochastic component of  $\ln(c_t)$  is neither

<sup>&</sup>lt;sup>3</sup>Notice that Lucas (1987) uses the unconditional mean operator instead of the conditional mean operator in (2.2). The same problem can be proposed using the conditional expectation instead. This is exactly how we proceed in this paper.

stationary nor ergodic. This led Obstfeld (1994) to use  $\mathbb{E}_0(\cdot)$  in defining welfare costs:

$$\mathbb{E}_{0}\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1+\lambda\right)c_{t}\right)=\sum_{t=0}^{\infty}\beta^{t}u\left(\mathbb{E}_{0}\left(c_{t}\right)\right).$$
(2.3)

Here,  $\lambda$  is the welfare cost associated with all the uncertainty in consumption, not just the uncertainty associated with the business-cycle component of consumption. Thus, it cannot be labelled the welfare cost of business cycles. Indeed, on an earlier paper (Issler, Franco, and Guillén (2008)), we have labelled it the *welfare cost of macroeconomic uncertainty* as opposed to the *welfare cost of business cycles*.

An interesting generalization of the setup in Lucas is due to Alvarez and Jermann (2004), who proposed offering the consumer a convex combination of  $\{c_t^*\}_{t=0}^{\infty}$  and  $\{c_t\}_{t=0}^{\infty}$ :  $(1 - \alpha) c_t + \alpha c_t^*$ , where  $c_t^* = \mathbb{E}_0(c_t)$ . They make the welfare cost to be a function of the weight  $\alpha$ ,  $\lambda(\alpha)$ , which solves:

$$\mathbb{E}_{0}\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1+\lambda\left(\alpha\right)\right)c_{t}\right) = \mathbb{E}_{0}\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1-\alpha\right)c_{t}+\alpha c_{t}^{*}\right).$$
(2.4)

In their setup  $\lambda(0) = 0$ , and  $\lambda$ , as defined by Lucas, is obtained as  $\lambda = \lambda(1)$ , when using  $\mathbb{E}(\cdot)$  instead of  $\mathbb{E}_0(\cdot)$  in (2.4). They label  $\lambda(1)$  as the *total cost of business* cycles and define the marginal cost of business cycles, obtained after differentiating (2.4) with respect to  $\alpha$  as<sup>4</sup>:

$$\lambda'(0) = \frac{\mathbb{E}_0 \sum_{t=0}^{\infty} \left[\beta^t u'(c_t) \times \mathbb{E}_0(c_t)\right]}{\mathbb{E}_0 \sum_{t=0}^{\infty} \left[\beta^t u'(c_t) \times c_t\right]} - 1.$$
(2.5)

As stressed by Alvarez and Jermann, there is a straightforward interpretation for  $\lambda'(0)$ . Consider a Taylor-expansion argument for  $\lambda(\alpha)$  around zero. We have:  $\lambda(\alpha) \cong \lambda(0) + \lambda'(0) \alpha$ . Recall that  $\lambda(0) = 0$ . Thus,  $\lambda(\alpha) \cong \lambda'(0) \alpha$ , which makes  $\lambda'(0)$  the first-order approximation of  $\lambda(1)$  around zero, recalling that  $\lambda(1)$ is Lucas' measure. Their setup relies solely on asset-pricing data to compute  $\lambda'(0)$ , which avoids completely the specification of preferences. However, as seen in (2.5), there is a preference counterpart of their formulas which will be used here as we show below.

 $<sup>^{4}</sup>$ We have to assume that the usual regularity conditions hold in exchanging the integral and derivative signs; see the conditions in Amemiya (1985, Theorem 1.3.2).

From (2.2), (2.3), and (2.5), notice that the total and the marginal cost of business cycles can be computed if consumption is stationary and ergodic and also when it is not. The only difference is whether we employ  $\mathbb{E}(\cdot)$  or  $\mathbb{E}_t(\cdot)$ , respectively, in defining it. Despite that, the choice of how to model consumption is an important one for several reasons. As is well known, unless consumption has a unit root, we cannot consider the existence of shocks with a permanent effect on it. The arguments in Reis (2009) in favor of consumption containing a unit root, what he has labelled Hall's (1978) consumption process, are convincing. Moreover, as stressed by Alvarez and Jermann (2005), "for many cases where the pricing kernel is a function of consumption, innovations to consumption need to have permanent effects." A permanent-transitory decomposition of consumption shocks allows to explicitly isolate transient and permanent sources of welfare fluctuations, which could, in principle, be associated with the welfare costs of business cycles and the welfare costs of growth components. If one does not separate the welfare costs associated with permanent and transitory components, there is the risk of inconsistent estimation of business-cycle costs alone.

As stressed in Issler and Vahid (2001), "theoretical models are rarely built in terms of permanent or transitory shocks. Rather, they are built in terms of real (e.g., productivity) or nominal (e.g., monetary) shocks." Here, in the original spirit of Lucas, we will link transitory shocks to sources of business cycles. Permanent shocks will be linked to sources of economic growth. Moreover, we impose independence between them. To go one step further would be to link these shocks, respectively, to monetary policy and to productivity, something we refrain from doing here. We rely on the argument put forth by Issler and Vahid who point or that not all "permanent" shocks are "productivity" shocks, since there may be permanent demand shocks to taste, for example. One could also think of transitory productivity shocks as well, challenging the link between "transitory" and "monetary." With that in mind, we now expose our own setup.

To start the discussion of difference-stationary consumption, we first assume that

the utility function is in CES class, with risk aversion coefficient  $\phi$ :

$$u(c_t) = \frac{c_t^{1-\phi} - 1}{1-\phi},$$
(2.6)

where  $u(c_t)$  approaches  $\ln(c_t)$  as  $\phi \to 1$ .

As shown in Beveridge and Nelson (1981), every linear difference-stationary process can be decomposed as the sum of a deterministic term, a random walk (martingale) trend, and a stationary cycle (ARMA process). The analogue of (2.1) when consumption is difference stationary is:

$$\ln(c_t) = \ln(\alpha_0) + \ln(1 + \alpha_1) \cdot t - \frac{\omega_t^2}{2} + \sum_{i=1}^t \varepsilon_i + \sum_{j=0}^{t-1} \psi_j \mu_{t-j}$$
(2.7)

where  $\ln \left[ \alpha_0 \left( 1 + \alpha_1 \right)^t \cdot \exp \left( -\omega_t^2 / 2 \right) \right]$  is the deterministic term,  $\sum_{i=1}^t \varepsilon_i$  is the random walk component,  $\sum_{j=0}^{t-1} \psi_j \mu_{t-j}$  is the  $MA(\cdot)$  representation of the stationary part (cycle), which entails  $\psi_0 = 1$  and  $\sum_{j=0}^{\infty} \psi_j^2 < \infty$ . The permanent shock  $\varepsilon_t$  and the transitory shock  $\mu_t$  are assumed to have a bivariate Normal distribution as follows:

$$\begin{pmatrix} \varepsilon_t \\ \mu_t \end{pmatrix} \sim i.i.d.\mathcal{N}\left(\begin{pmatrix} 0 \\ 0 \end{pmatrix}, \begin{pmatrix} \sigma_{11} & 0 \\ 0 & \sigma_{22} \end{pmatrix}\right), \qquad (2.8)$$

i.e., shocks are uncorrelated across time and are contemporaneously uncorrelated. This implies independence across time for both shocks and independence among them too. Thus,  $\omega_t^2 = \sigma_{11} \cdot t + \sigma_{22} \sum_{j=0}^{t-1} \psi_j^2$  is the conditional variance of  $\ln(c_t)$ , where it becomes clear that  $\varepsilon_t$  and  $\mu_t$  have two very different roles in terms of uncertainty: the uncertainty of  $\varepsilon_t$  grows without bound with t ( $\sigma_{11} \cdot t$ ), whereas that of  $\mu_t$  also increases with  $t \left(\sigma_{22} \sum_{j=0}^{t-1} \psi_j^2\right)$  but is bounded from above by the unconditional variance  $\sigma_{22} \sum_{j=0}^{\infty} \psi_j^2$ .

As noted by Reis (2009), the degree of persistence imposed in the process  $\{\ln(c_t)\}_{t=1}^{\infty}$ is critical to determine the welfare costs of business cycles. As an example, suppose we use a first-order autoregressive AR(1) assumption for  $\ln(c_t)$  about a deterministic trend, i.e.,  $\ln(c_t) = \ln(\alpha_0) + \ln(1 + \alpha_1) \cdot t - \frac{\omega_t^2}{2} + \sum_{j=0}^{\infty} \psi^j \mu_{t-j}$ , where  $\omega_t^2 = \sigma_{22} \sum_{j=0}^{t-1} \psi^{2j}$ and  $\psi$  is the first-order autoregressive coefficient, with  $|\psi| < 1$ . Then, the variance of  $\ln(c_t)$  about its trend is  $\frac{\sigma_{22}}{1-\psi^2}$ . Making  $\{\ln(c_t)\}_{t=1}^{\infty}$  more persistent implies letting  $|\psi|$  approach unity from below and  $\frac{\sigma_{22}}{1-\psi^2}$  to grow without bound. Since the consumer dislikes risk, the welfare cost of business cycles is an increasing function of the persistence in  $\{\ln(c_t)\}_{t=1}^{\infty}$ .

Additionally, there is a discontinuity of the asymptotics for the least-square estimate of  $\psi$ ,  $\hat{\psi}$ , at  $|\psi| = 1$ , when one uses a sample size of T observations in estimation. If  $|\psi| < 1$ ,  $\hat{\psi}$  is  $\sqrt{T}$ -consistent, whereas, at  $\psi = 1$ , it is T-consistent and downward biased in small samples<sup>5</sup>. Reis applies two alternative methods to compute welfare costs if consumption has a unit root. The first is a local-to-unity approach, where the unit root only shows up in the limit. Alternatively, based on the results of several tests, Reis also imposes a unit root to consumption, avoiding the downward-bias problem in estimation. As can be seen from equation (2.7), we chose to impose a unit root to consumption as well<sup>6</sup>. However, we go one step further since we separate the welfare effects of permanent and transitory shocks to  $\ln(c_t)$  given the structure underlying (2.8).

A main objective of this paper is to isolate the welfare costs of business cycles and the welfare costs of economic growth. As stressed by Issler, Franco, and Guillén (2008), one way to study the welfare cost of business cycles in a difference-stationary world is to work with independent shocks responsible for trend and cyclical movements in  $\ln (c_t)$ . If one does not separate the effects of these shocks, she/he is forced to examine the welfare cost of *all* macroeconomic uncertainty, or to work with a tainted measure of welfare cost of business cycles which encompasses some or all of the cost associated with economic-growth factors. A previous attempt to deal with this issue includes only examining consumption fluctuations at business-cycle horizons; see, e.g., Alvarez and Jermann (2004). In our view, this strategy is best viewed as an approximation, since some business-cycle variation in consumption can be due to permanent shocks: recall that one of the main features of the real-business-cycle

<sup>&</sup>lt;sup>5</sup>Moreover, the effect of uncertainty is very different for welfare costs. As  $\psi \to 1$ , the autoregressive process becomes a random walk, for which the conditional variance is  $\sigma_{22} \cdot t$ , i.e., increases without bound with time.

<sup>&</sup>lt;sup>6</sup>Reis claims that "Consumption growth is positively serially correlated, a fact that has inspired most modern research on consumption." Indeed, the models we entretain below have this character.

literature was that permanent shocks could indeed generate business-cycle fluctuations; see, *inter alia*, Kydland and Prescott (1982), King, Plosser and Rebelo (1987), King, Plosser, Stock and Watson (1991), and Issler and Vahid (2001).

In the framework above, because of independence of shocks, it is natural to evaluate the welfare cost of business cycles using  $\mu_t$ , and to evaluate the welfare cost of economic growth using  $\varepsilon_t$ . To do so, consider the two processes below, where we start with (2.7) and shut out permanent and transitory shocks, respectively, as follows:

$$c_t^T = \alpha_0 \left(1 + \alpha_1\right)^t \cdot \exp\left[-\frac{\sigma_{22} \sum_{j=0}^{t-1} \psi_j^2}{2}\right] \sum_{j=0}^{t-1} \psi_j \mu_{t-j}, \text{ and},$$
(2.9)

$$c_t^P = \alpha_0 \left(1 + \alpha_1\right)^t \cdot \exp\left(-\frac{\sigma_{11}}{2}t\right) \sum_{i=1}^t \varepsilon_i.$$
(2.10)

From (2.7), we can think of  $c_t^T$  and  $c_t^P$  as limit cases, respectively:

$$\lim_{\sigma_{11}\to 0} c_t = c_t^T, \text{ and } \lim_{\sigma_{22}\to 0} c_t = c_t^P.$$

We propose measuring the welfare cost for the representative consumer of bearing the uncertainty associated with  $\{\mu_t\}$  alone (business cycles) through the use of  $c_t^P$ . Notice that the conditional means of  $c_t^P$  and  $c_t$  are identical:  $\mathbb{E}_0(c_t^P) = \mathbb{E}_0(c_t) = \alpha_0(1+\alpha_1)^t$ . However, the uncertainty of the consumption stream  $\{c_t\}_{t=1}^{\infty}$  is larger than that of  $\{c_t^P\}_{t=1}^{\infty}$ . Thus,  $c_t$  is a mean-preserving spread of  $c_t^P$ . Risk averse consumers prefer the stream  $\{c_t^P\}_{t=1}^{\infty}$  over  $\{c_t\}_{t=1}^{\infty}$ . Thus, we measure the welfare cost associated with  $\{\mu_t\}$  alone using  $\lambda_P$ , which solves:

$$\mathbb{E}_0\left[\sum_{t=0}^{\infty}\beta^t u\left(\left(1+\lambda_P\right)c_t\right)\right] = \mathbb{E}_0\left[\sum_{t=0}^{\infty}\beta^t u\left(c_t^P\right)\right],\tag{2.11}$$

i.e., we can think of  $\lambda_P$  as the welfare cost of bearing the risks associated with transitory shocks alone. Thus, we label it the welfare cost of business cycles.

In order to implement the computation of  $\lambda_P$ , we specialize the utility function to in the CES class as in (2.6). After straightforward but tedious algebra we get,

$$\lambda_P = \exp\left(\phi \widetilde{\sigma}_{22}/2\right) - 1, \qquad (2.12)$$

where, for the sake of simplicity in computation, we replace  $\sigma_{22} \sum_{j=0}^{t-1} \psi_j^2$  by its respective unconditional counterpart  $\tilde{\sigma}_{22} = \sigma_{22} \sum_{j=0}^{\infty} \psi_j^2$ . We also assume that the convergence condition  $\beta \cdot (1 + \alpha_1)^{(1-\phi)} \cdot \exp^{(-\phi(1-\phi)\sigma_{11}/2)} < 1$  holds. Notice that the welfare cost of business cycles does not depend on the uncertainty associated with permanent shocks. However, it depends on  $\sigma_{22}$  – the uncertainty behind transitory shocks – as well as on the degree of persistence of these shocks, captured by  $\sum_{j=0}^{\infty} \psi_j^2$ , and on the relative risk-aversion coefficient  $\phi$ .

Analogously, we propose measuring the welfare cost for the representative consumer of bearing the uncertainty associated with  $\{\varepsilon_t\}$  alone (economic growth) through the use of  $c_t^T$ . Recall that  $\mathbb{E}_0(c_t^T) = \mathbb{E}_0(c_t) = \alpha_0(1+\alpha_1)^t$ , and  $c_t$  is a mean-preserving spread of  $c_t^T$ . Hence, we measure the welfare cost associated with  $\{\varepsilon_t\}$  alone by using  $\lambda_T$ , which solves:

$$\mathbb{E}_{0}\left[\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1+\lambda_{T}\right)c_{t}\right)\right] = \mathbb{E}_{0}\left[\sum_{t=0}^{\infty}\beta^{t}u\left(c_{t}^{T}\right)\right]$$
(2.13)

Hence, we can think of  $\lambda_T$  as the welfare cost of economic growth. Using (2.6), one can show that:

$$\lambda_T = \begin{cases} \left[ \frac{\left(1 - \beta \cdot (1 + \alpha_1)^{(1 - \phi)} \cdot \exp(-\phi(1 - \phi)\sigma_{11}/2)\right)}{\left(1 - \beta \cdot (1 + \alpha_1)^{(1 - \phi)}\right)} \right]^{\frac{1}{(1 - \phi)}} - 1, & \text{for } \phi \neq 1 \\ \\ \exp\left(\frac{\beta\sigma_{11}}{2(1 - \beta)}\right) - 1, & \text{for } \phi = 1 \end{cases}$$

$$(2.14)$$

where we assume that the convergence condition  $\beta \cdot (1 + \alpha_1)^{(1-\phi)} < 1$ , holds. Notice that  $\lambda_T$  does not depend on  $\sigma_{22}$  – i.e., on how uncertain transitory shocks are. However, it depends on  $\beta$ ,  $\phi$ ,  $\sigma_{11}$  and  $\alpha_1$ .

Finally, we can compute welfare costs for the representative consumer of bearing the uncertainty associated with both  $\{\varepsilon_t\}$  and  $\{\mu_t\}$  by introducing  $c_t^D$ :

$$\lim_{\sigma_{11} \to 0, \sigma_{22} \to 0} c_t = c_t^D = \alpha_0 \left( 1 + \alpha_1 \right)^t.$$
(2.15)

Here,  $\mathbb{E}_0(c_t^D) = \mathbb{E}_0(c_t) = \alpha_0(1+\alpha_1)^t$ , making  $c_t$  a mean-preserving spread of  $c_t^D$ . We measure the welfare cost associated with both  $\{\varepsilon_t\}$  and  $\{\mu_t\}$  using  $\lambda_D$ , which solves:

$$\mathbb{E}_{0}\left[\sum_{t=0}^{\infty}\beta^{t}u\left(\left(1+\lambda_{D}\right)c_{t}\right)\right] = \mathbb{E}_{0}\left[\sum_{t=0}^{\infty}\beta^{t}u\left(c_{t}^{T}\right)\right].$$
(2.16)

Using (2.6), we obtain:

$$\lambda_D = \begin{cases} \left[ \frac{e^{\phi(1-\phi)\tilde{\sigma}_{22}/2} \cdot \left(1-\beta \cdot (1+\alpha_1)^{(1-\phi)} \cdot e^{(-\phi(1-\phi)\sigma_{11}/2)}\right)}{\left(1-\beta \cdot (1+\alpha_1)^{(1-\phi)}\right)} \right]^{\frac{1}{(1-\phi)}} - 1, & \text{for } \phi \neq 1 \\ e^{\frac{\beta\sigma_{11}+(1-\beta)\tilde{\sigma}_{22}}{2(1-\beta)}} - 1, & \text{for } \phi = 1 \end{cases}$$

$$(2.17)$$

where we assume that the convergence condition  $\beta \cdot (1 + \alpha_1)^{(1-\phi)} < 1$ , holds.

Measures  $\lambda_P$ ,  $\lambda_T$ , and  $\lambda_D$  are what Alvarez and Jermann have labelled measures of *total welfare costs*. Here, we are also interested in measures of *marginal welfare costs*, i.e.,  $\lambda'_P(0)$ ,  $\lambda'_T(0)$ , and  $\lambda'_D(0)$ . Starting from (2.4), and using (2.6), we measure marginal welfare costs of business cycles, economic growth, and macroeconomic uncertainty by using  $c_t^P$ ,  $c_t^T$ , and  $c_t^D$ , respectively:

$$\lambda_{P}'(0) = \exp\left(\phi\widetilde{\sigma}_{22}\right) - 1, \qquad (2.18)$$

$$\lambda_{T}'(0) = \begin{cases} \frac{\left(1 - \beta \cdot (1 + \alpha_{1})^{(1 - \phi)} \cdot e^{\frac{-\phi(1 - \phi)\sigma_{11}}{2}}\right)}{\left(1 - \beta \cdot (1 + \alpha_{1})^{(1 - \phi)} \cdot e^{\frac{\phi(1 + \phi)\sigma_{11}}{2}}\right)} - 1, \text{ for } \phi \neq 1 \\ , \text{ and,} \qquad (2.19) \end{cases}$$

$$\lambda_{D}'(0) = \begin{cases} \frac{1-\beta}{1-\beta e^{\sigma_{11}}} - 1, & \text{for } \phi = 1\\ \frac{e^{\phi \cdot \tilde{\sigma}_{22}} \cdot \left(1-\beta \cdot (1+\alpha_{1})^{(1-\phi)} \cdot e^{\frac{-\phi(1-\phi)\sigma_{11}}{2}}\right)}{\left(1-\beta \cdot (1+\alpha_{1})^{(1-\phi)} \cdot e^{\frac{\phi(1+\phi)\sigma_{11}}{2}}\right)} - 1, & \text{for } \phi \neq 1\\ \frac{e^{\tilde{\sigma}_{22}}(1-\beta)}{1-\beta e^{\sigma_{11}}} - 1, & \text{for } \phi = 1 \end{cases}$$
(2.20)

where we assume that the usual specific convergence conditions apply in computing  $\lambda'_P(0)$ ,  $\lambda'_C(0)$  and  $\lambda'_T(0)$ , respectively, and replace  $\sigma_{22} \sum_{j=0}^{t-1} \psi_j^2$  by its respective unconditional counterpart  $\tilde{\sigma}_{22} = \sigma_{22} \sum_{j=0}^{\infty} \psi_j^2$  in computing  $\lambda'_C(0)$ . As in the case of total welfare costs, we interpret  $\lambda'_P(0)$ ,  $\lambda'_T(0)$ , and  $\lambda'_D(0)$  as being the marginal welfare costs of business cycles, of economic growth, and of all macroeconomic uncertainty, respectively.

Finally, we give some intuition behind the measures of welfare costs proposed

above. One way to think about (2.10) is:

$$\lim_{\sigma_{22}\to 0} c_t = c_t^P = \mathbb{E}\left[c_t \mid \mathcal{I}_0, \{\varepsilon_t\}_{t=0}^{\infty}\right]$$

which shows that  $c_t^P$  is the conditional expectation of  $c_t$  when we have *perfect fore-sight* of the sequence  $\{\varepsilon_t\}_{t=0}^{\infty}$  of permanent shocks. Thus, in computing the welfare costs of business cycles, we *control* for the existence of permanent shocks to consumption. This shows that the welfare-cost measures  $\lambda_P$  and  $\lambda'_P(0)$  only take into account the uncertainty that goes beyond permanent shocks, i.e., transitory shocks alone.

Using (2.9) and (2.15), a similar reasoning applies to  $c_t^T$  and  $c_t^D$ , respectively:

$$\lim_{\sigma_{11}\to 0} c_t = c_t^T = \mathbb{E}\left[c_t \mid \mathcal{I}_0, \{\mu_t\}_{t=0}^{\infty}\right],$$
$$\lim_{\sigma_{11}\to 0, \sigma_{22}\to 0} c_t = c_t^D = \mathbb{E}\left[c_t \mid \mathcal{I}_0, \{\mu_t\}_{t=0}^{\infty}, \{\varepsilon_t\}_{t=0}^{\infty}\right]$$

## 3. Identification and Estimation of Structural Parameters used in Computing $\lambda_T$ , $\lambda_P$ , $\lambda_D$ , $\lambda'_T(0)$ , $\lambda'_P(0)$ , and $\lambda'_D(0)$

Next, we discuss the reduced form and the structural form used in estimating  $\lambda_T$ ,  $\lambda_P$ ,  $\lambda_D$ ,  $\lambda'_T(0)$ ,  $\lambda'_P(0)$ , and  $\lambda'_D(0)$ . For the reduced form, we borrow heavily from the discussion in Issler, Franco, and Guillén (2008). This is especially important regarding possible long-run constraints in the data. Our starting point is a vector autoregression (VAR), where possible cointegrating restrictions are used in estimation. We show how a simple identification strategy can be used in this setup, although it does not impose the restriction that  $\mathbb{E}(\varepsilon_t \mu_t) = 0$ . For that reason, we also discuss structural time-series models based on Harvey (1985b) and Koopman et al. (2009) where  $\mathbb{E}(\varepsilon_t \mu_t) = 0$  is imposed under joint Normality for shocks.

#### 3.1. Reduced Form: Long-Run and Short-Run Constraints

A full discussion of the econometric models employed here can be found in Beveridge and Nelson (1981), Stock and Watson (1988), Engle and Granger (1987), Campbell (1987), Campbell and Deaton (1989), Vahid and Engle (1993), and Proietti (1997). Denote by  $y_t = (\ln (c_t), \ln (I_t))'$  a 2×1 vector containing respectively the logarithms of consumption and disposable income per-capita. We assume that both series contain a unit-root and are possibly cointegrated as in  $[-1, 1]' y_t$  because of the Permanent-Income Hypothesis (Campbell(1987)). A vector error-correction model (VECM(p-1)) is:

$$\Delta y_{t} = \Gamma_{1} \Delta y_{t-1} + \ldots + \Gamma_{p-1} \Delta y_{t-p+1} + \gamma [-1, 1] y_{t-p} + \eta_{t}.$$
(3.1)

Here, long-run constraints in the VAR are imposed through the error-correction mechanism  $[-1,1] y_{t-p}$ . As discussed in Vahid and Engle (1993), short-run restrictions in the form of common cycles can be imposed in (3.1). Let  $[-1,1]' = \alpha$  and consider the following restrictions on the parameters of (3.1):

$$\widetilde{\alpha}' \Gamma_i = 0$$
, for all  $i = 1, 2, \cdots p - 1$ , and  $\widetilde{\alpha}' \gamma = 0$ .

Then, we can represent (3.1) as having *common-cyclical-feature* restrictions in a  $2 \times 1$  system:

$$\begin{bmatrix} 1 & \widetilde{\alpha}^{*'} \\ \mathbf{0} & 1 \end{bmatrix} \Delta y_t = \begin{bmatrix} 0 & \cdots & 0 & 0 \\ A_1^* & \cdots & A_{p-1}^* & \gamma^* \end{bmatrix} \begin{bmatrix} \Delta y_{t-1} \\ \vdots \\ \Delta y_{t-p+1} \\ \alpha' y_{t-p} \end{bmatrix} + \eta_t, \quad (3.2)$$

where  $A_1^*, \dots, A_{p-1}^*, \gamma^*$  represent partitions of  $\Gamma_1, \dots, \Gamma_{p-1}, \gamma$ , respectively. Notice that  $\begin{bmatrix} 1 & \widetilde{\alpha}^{*'} \\ \mathbf{0} & 1 \end{bmatrix}$  is non-singular, which allows to recover the parameters in (3.1) from the ones in (3.2) as we pre-multiply the latter by  $\begin{bmatrix} 1 & \widetilde{\alpha}^{*'} \\ \mathbf{0} & 1 \end{bmatrix}^{-1}$ . Indeed, (3.2) is just a more parsimonious representation than (3.1).

#### 3.2. Structural Time-Series Models with Long-Run Constraints

Regarding our purposes here, the main problem of the reduced-form approach described in the previous section is that it does not impose the constraint that permanent and transitory shocks to  $\ln(c_t)$  are orthogonal. Under Normality, this would imply independence of these shocks. For that reason, we now turn to the discussion of *structural* time-series models, where possible long- and short-run restrictions are still kept in a different setup. Here, we present a brief summary of the *structural* time-series model of Harvey (1985b) and Koopman et al. (2009). We start the discussion using a univariate framework. There, the main objective is to decompose a single integrated series (I(1)) in a trend and a cycle, treating both as latent variables to be estimated by maximum likelihood, which guarantees consistent and asymptotically Normal parameter estimates, a key property in our case.

For a single economic series  $x_t$ , we decompose it as:

$$x_t = \tau_t + \varphi_t$$

where  $\tau_t$  is the I(1) trend,  $\varphi_t$  is the cycle. Shocks to each of these two components are independent of each other and also across time. The trend evolves as:

$$\tau_t = \tau_{t-1} + \delta + \upsilon_t, \tag{3.3}$$

where  $v_t$  has variance given by  $\sigma_v^2$ , whereas the cyclical component evolves as a bivariate VAR(1):

$$\begin{bmatrix} \varphi_t \\ \varphi_t^* \end{bmatrix} = \rho \begin{bmatrix} \cos \lambda & \sin \lambda \\ -\sin \lambda & \cos \lambda \end{bmatrix} \begin{bmatrix} \varphi_{t-1} \\ \varphi_{t-1}^* \end{bmatrix} + \begin{bmatrix} \omega_t \\ \omega_t^* \end{bmatrix}$$
(3.4)

where the component  $\varphi_t^*$  shows up by construction; see Harrison and Akran (1983). Both  $\omega_t$  and  $\omega_t^*$  are orthogonal white noise errors with variances given by  $\sigma_{\omega}^2$  and  $\sigma_{\omega}^{*2}$ , respectively. Harvey (1985b) argues that very little is lost in terms in terms of fit if we impose the restriction that  $\sigma_{\omega}^2 = \sigma_{\omega}^{*2}$ , representing an advantage in terms of parsimony. Finally, some restrictions on parameter values should be observed:

$$0 \le \lambda^* \le \pi$$
 and  $0 < \rho \le 1$ ,

where  $\lambda^*$  is the frequency of the cycle and  $\rho$  is the discount factor for its amplitude. The last restriction makes the cyclical component stationary.

One can also show that the cyclical component obeys:

$$\varphi_t = \frac{\left(1 - \rho \cos \lambda L\right)\omega_t + \left(\rho \sin \lambda L\right)\omega_t^*}{1 - 2\rho \cos \lambda L + \rho^2 L^2}$$

where L is the lag operator,  $L^k x_t = x_{t-k}$ . Under  $\sigma_{\omega}^2 = \sigma_{\omega}^{*2}$ , we can put the last equation in an ARMA format as:

$$\left(\rho^{2}L^{2} - 2\rho\cos\lambda L + 1\right)\varphi_{t} = \left(1 + \Phi L\right)\omega_{t}$$

where it becomes clear that  $\varphi_t$  follows an ARMA(2, 1), with  $\Phi = \rho(\sin \lambda - \cos \lambda)$ . This is a restriction into the ARMA class of models, since not every cycle of an economic series will be well modelled as an ARMA(2, 1).

Following the notation for the univariate class of models, in a multivariate setting, we can represent  $y_t = (\ln(c_t), \ln(I_t))'$  as having a common trend and a common cycle, respectively, as:

$$\begin{bmatrix} \ln (c_t) \\ \ln (I_t) \end{bmatrix} = \begin{bmatrix} 1 \\ 1 \end{bmatrix} \tau_t + \begin{bmatrix} 1 \\ \theta \end{bmatrix} \varphi_t, \qquad (3.5)$$

where the I(1) trend component  $\tau_t$  follows (3.3) and the stationary cyclical component  $\varphi_t$  follows (3.4). Here, the bivariate system in  $y_t$  is modelled with just a single stochastic trend and a single cycle, respectively. The trend affects identically the two series in  $y_t$ , whereas the cycle affects them differently. The vector [-1, 1]'removes the common trend and that the vector  $[-\theta, 1]'$ removes the common cycle, where there is the additional restriction that  $\theta \neq 1^7$ . The structural time-series model in (3.5) is analogous to its reduced-form counterpart (3.2), in which it imposes identical long- and short-run restrictions. Despite that, they differ in which (3.5) imposes independence for the shocks to  $\tau_t$  and  $\varphi_t$ , whereas (3.2) does not.

As stressed by Issler and Vahid (2001), there are several theoretical reasons why consumption and income should cointegrate (Campbell (1987)) and have common cycles (King, Plosser, and Rebelo (1988), Campbell and Mankiw (1989), and King et al. (1991)). Despite that, one may be more willing to impose long-run restrictions than short-run restrictions, meaning that the two variables in  $y_t$  have two distinct cycles<sup>8</sup>, but still a common trend as in (3.5). This can be easily accommodated by

<sup>&</sup>lt;sup>7</sup>Testing for common cycles in a multivarite framework is discussed by Carvalho, Harvey, and Trimbur (2007).

<sup>&</sup>lt;sup>8</sup>See the discussion and proposed tests in Carvalho, Harvey, and Trimbur (2007).

the structure in (3.5), where  $\begin{bmatrix} 1\\ \theta \end{bmatrix} \varphi_t$  is replaced by the 2 × 1 vector  $\varphi_t$ ,

$$\begin{bmatrix} \varphi_t \\ \varphi_t^* \end{bmatrix} = \begin{bmatrix} \rho \begin{pmatrix} \cos \lambda & \sin \lambda \\ -\sin \lambda & \cos \lambda \end{pmatrix} \otimes \mathbf{I}_2 \end{bmatrix} \begin{bmatrix} \varphi_{t-1} \\ \varphi_{t-1}^* \end{bmatrix} + \begin{bmatrix} \omega_t \\ \omega_t^* \end{bmatrix}, \quad (3.6)$$

where now  $\varphi_t$ ,  $\varphi_t^*$ ,  $\omega_t$  and  $\omega_t^*$  are 2 × 1 vectors,  $\lambda$  is a 1 × 2 vector, and we impose the restriction that  $\mathbb{E}(\omega_t \omega_t') = \mathbb{E}(\omega_t^* \omega_t^{*'}) = \Sigma_{\omega}$ , making  $\operatorname{VAR}\begin{bmatrix}\omega_t\\\omega_t^*\end{bmatrix} = \mathbf{I}_2 \otimes \Sigma_{\omega}$ .

The univariate and multivariate models discussed above can be easily put in state-space form with Normal disturbances, where the Kalman Filter can be used to compute the likelihood function through the one-step prediction error decomposition. Consistent and asymptotically Normal estimates of parameter values are thus obtained, which is a critical step to construct our estimates of  $\lambda_T$ ,  $\lambda_P$ ,  $\lambda_D$ ,  $\lambda'_T(0)$ ,  $\lambda'_P(0)$ , and  $\lambda'_D(0)$ , as well as to construct their respective asymptotic confidence intervals; more details on state-space forms, the likelihood function, and the use of the Kalman Filter can be found in Koopman et al. (2009, Chapter 9).

Finally, we discuss the identification of the key parameters in the welfare-cost formulas of Section 2 by using  $\tau_t$  and  $\varphi_t$ : the variances  $\sigma_{11}$  and  $\tilde{\sigma}_{22}$  and the instantaneous growth rate of consumption,  $\alpha_1$ . The parameter  $\sigma_{11}$  can be identified using VAR( $\Delta \tau_t$ ), whereas  $\tilde{\sigma}_{22} = \sigma_{22} \sum_{j=0}^{\infty} \psi_j^2$  can be identified by using VAR( $\varphi_t$ ). If one uses the model with a common trend, but idiosyncratic cycles as in (3.6), identification of  $\tilde{\sigma}_{22}$  is still straightforward by using VAR( $[1, 0] \times \varphi_t$ ). It is easy to identify  $\ln (1 + \alpha_1)$ employing  $\mathbb{E} (\Delta \tau_t)$ .

The identification strategy outlined above suggests how to estimate consistently  $\alpha_1$ ,  $\sigma_{11}$ , and  $\tilde{\sigma}_{22}$ , as well as how to compute the variances of these estimates. These are based on Phillips and Solo (1992), who discuss how to compute consistent estimates of parameters of linear processes transformed using the Beveridge and Nelson (1981) filter. First, running a regression of  $\Delta \tau_t$  on a constant provides a consistent estimate of  $\ln(1 + \alpha_1)$ :  $\widehat{\ln(1 + \alpha_1)} = \frac{1}{T} \sum_{t=1}^{T} \Delta \tau_t$ , where T is the sample size used in estimation. Using Slutsky's Theorem, it is straightforward to find a consistent estimate for  $\alpha_1$ . Since the cycle is a zero-mean stationary and ergodic linear process with serial dependence,  $\widehat{\sigma}_{22} = \frac{1}{T} \sum_{t=1}^{T} \varphi_t^2$  is a consistent estimate of  $\widetilde{\sigma}_{22}$ . On the other

hand, the first difference of the trend,  $\Delta \tau_t$ , is still a linear process, but serially independent. Hence,  $\widehat{\sigma_{11}} = \frac{1}{T} \sum_{t=1}^{T} \left( \Delta \tau_t - \ln \widehat{(1 + \alpha_1)} \right)^2$  is a consistent estimate of  $\sigma_{11}$ .

As long as the serial dependence is not too strong – as is the case for all estimates above – it poses no problem to estimate consistently  $\alpha_1$ ,  $\sigma_{11}$ , and  $\tilde{\sigma}_{22}$ . But we must account properly for the existence of serial dependence in order to estimate consistently the variance of  $\widehat{\alpha_1}$ ,  $\widehat{\sigma_{11}}$ , and  $\widehat{\widetilde{\sigma}_{22}}$ , which are all sample means. In our context, if the elements in these sample means have serial dependence and heterogeneity of unknown form, their variances can still be consistently estimated using the concept of long-run variance, which is given by  $\gamma_0 + 2 \cdot \sum_{i=1}^{\infty} \gamma_i$ , where  $\gamma_i$  is the *i*-th auto-covariance of the terms in the sample mean<sup>9</sup>. Based on the fact that  $\sqrt{T}\left(\widehat{\sigma_{11}} - \sigma_{11}\right) \xrightarrow{d} \mathcal{N}(0, V_{11}), \ \sqrt{T}\left(\widehat{\widetilde{\sigma}_{22}} - \widetilde{\sigma}_{22}\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \text{and} \ \sqrt{T}\left(\widehat{\alpha_1} - \alpha_1\right) \xrightarrow{d} \mathcal{N}(0, V_{22}), \ \mathbb{N}(0, V_{22}), \ \mathbb{N}(0,$  $\mathcal{N}(0, V_{\alpha})$  it is straightforward to estimate consistently  $V_{11}, V_{22}$  and  $V_{\alpha}$ . In our context, the only sample mean for which the elements are serially dependent is  $\frac{1}{T} \sum_{i=1}^{T} \varphi_t^2$ , whereas those in  $\frac{1}{T} \sum_{t=1}^{T} \Delta \tau_t$  and  $\frac{1}{T} \sum_{t=1}^{T} \left( \Delta \tau_t - \ln (1 + \alpha_1) \right)^2$  are independent. Implementing a long-run-variance estimate for  $V_{22}$  can be easily accomplished by using Newey and West's (1987) non-parametric procedure, which relies on consistent estimates of the auto-covariances of  $\varphi_t^2$  and a truncation window for computing a weighted average of them using a Bartlett kernel.

# **3.3.** Computing Asymptotic Confidence Intervals for Welfare Costs $\lambda_T$ , $\lambda_P$ , $\lambda_D$ , $\lambda'_T(0)$ , $\lambda'_P(0)$ , and $\lambda'_D(0)$

In this section, we show how to compute asymptotic confidence intervals for welfarecost estimates based on (2.12), (2.14), (2.18), (2.19), and (2.20). As discussed in the previous section, we are able to identify  $\sigma_{11}$ ,  $\tilde{\sigma}_{22}$ , and  $\alpha_1$ , based on consistent and asymptotically Normal estimates (maximum likelihood) obtained for the unobserved-

<sup>&</sup>lt;sup>9</sup>For any sample average  $\frac{1}{T} \sum_{t=1}^{T} x_t$ , of satationary and ergodic linear series  $x_t$ , serially dependent, the long-run variance is  $\gamma_0 + 2 \cdot \sum_{i=1}^{\infty} \gamma_i$ , where  $\gamma_i$  is the *i*-th auto-covariance of  $x_t$ , i.e.,  $\gamma_i = \mathbb{E}\left[(x_t - \mu) (x_{t-i} - \mu)\right] = \mathbb{E}\left[(x_t - \mu) (x_{t+i} - \mu)\right].$ 

component model proposed by Harvey (1985b) and Koopman et al. (2009). Given these estimates, asymptotic confidence intervals can be obtained using the Delta Method.

Consider first the set of parameters  $\theta^* = (\beta, \phi, \sigma_{11}, \tilde{\sigma}_{22}, \alpha_1)'$ . All welfare costs  $\lambda_T$ ,  $\lambda_P, \lambda_D, \lambda'_T(0), \lambda'_P(0)$ , and  $\lambda'_D(0)$  can be expressed as specific non-linear functions of  $\theta^*$ . Here, we follow the literature in treating  $\beta$  and  $\phi$  as known (fixed), whereas the remaining parameters in  $\theta^*$ , stacked in  $\theta = (\sigma_{11}, \tilde{\sigma}_{22}, \alpha_1)'$ , are estimated consistently employing a sufficiently large sample of  $t = 1, 2, \dots, T$  observations. In this setup, the uncertainty in estimating  $\lambda_T, \lambda_P, \lambda_D, \lambda'_T(0), \lambda'_P(0)$ , and  $\lambda'_D(0)$  will be a function of the uncertainty in estimating the components of  $\theta$  alone, and the Delta Method can be used to compute asymptotic standard errors (and asymptotic confidence intervals) for welfare-cost estimates.

Suppose that a generic welfare measure  $\lambda^*$  relates to  $\theta$  as:

$$\lambda^* = G\left(\theta\right),\,$$

where  $G(\theta)$  is a continuous and continuously differentiable function. Here, the function  $G(\cdot)$  is specific to each welfare cost in equations (2.12), (2.14), (2.18), (2.19), and (2.20), and it can be verified that all the assumptions required to use the Delta Method are valid, case by case.

Given that a Central Limit Theorem holds for  $\theta$ ,

$$\sqrt{T}\left(\widehat{\theta}-\theta\right)\stackrel{d}{\to}\mathcal{N}\left(0,V\right),$$

the Delta Method can be employed to compute asymptotic confidence intervals for  $\widehat{\lambda^*}$ , which are based on:

$$\sqrt{T}\left(\widehat{\lambda^*} - \lambda^*\right) \xrightarrow{d} \mathcal{N}\left(0, \frac{\partial G\left(\theta\right)}{\partial \theta} V \frac{\partial G\left(\theta\right)}{\partial \theta'}\right).$$

In practice, we have to replace V with a consistent estimator,  $\hat{V}$ , and evaluate  $\frac{\partial G(\theta)}{\partial \theta}$ and  $\frac{\partial G(\theta)}{\partial \theta'}$  at  $\theta = \hat{\theta}$ . In this context, the estimated variance of  $\hat{\lambda^*}$  in finite samples is given by  $\frac{1}{T} \frac{\partial G(\theta)}{\partial \theta} \Big|_{\theta = \hat{\theta}} \hat{V} \frac{\partial G(\theta)}{\partial \theta'} \Big|_{\theta = \hat{\theta}}$ , and the 95% confidence interval for testing  $H_0$ :  $\lambda^* = 0$ , is given by  $\hat{\lambda^*} \pm 1.96 \times \sqrt{\frac{1}{T} \frac{\partial G(\theta)}{\partial \theta}} \Big|_{\theta = \hat{\theta}} \hat{V} \frac{\partial G(\theta)}{\partial \theta'} \Big|_{\theta = \hat{\theta}}$ .

#### 4. Empirical Results

Data for consumption of non-durables and services were obtained from DRI from 1929 through 2010. Data for consumption of perishables and services from 1901 to 1929 were obtained from Kuznets (1961) in real terms, and then chained with DRI data, resulting in a long-span series for consumption of non-durables and services from 1901-2011. Data for real GNP were also extracted from DRI from 1929 through 2010 and from Kuznets from 1901 through 1929. Data on population were extracted from Kuznets and DRI, and then chained. Figure 1 presents the data on consumption and income per-capita for the whole period 1901-2010. The peculiar features are first the magnitude of the great depression in both consumption and income behavior, and second the fact that pre-WWII data present much more volatility than post-WWII data.

We fitted a bivariate vector autoregression for the logs of consumption and income. Lag length selection indicated that a VAR(2) with an unrestricted constant term was an appropriate description of the dynamic system. This was true not only in terms of minimizing information criteria but also because this specification did not fail diagnostic testing.

Table 1 presents results of the cointegration test using Johansen's (1988, 1991) technique. The *Trace Statistics* for the null of no cointegration and of at most one cointegrating vector were respectively 16.43 and 0.18. At 5% significance, we conclude that there is one cointegrating vector, which estimate is given by (-1.000, 1.005)'. Conditioning on the existence of one cointegrating vector, we tested the restriction that it was equal to (-1, 1)'. We used the likelihood-ratio test in Johansen (1991), which yields a p-value of 0.831, not rejecting the null at usual levels of significance. An interesting by-product of cointegration analysis is testing the significance of the error-correction term in each regression of the system. The *t-statistic* associated with this test are -0.07 and 3.16, for the regression involving consumption and income respectively. Hence, the error-correction term affects income but not consumption, and the latter is long-run *weakly exogenous* in the sense of Engle, Hendry, and Richard (1983) and Johansen (1992). Despite that, we find Granger (1969) causality from income to consumption – the coefficient of lagged income is significant in consumption's equation – a t-statistic of -3.67. This shows the usefulness of the bivariate setup employed here, since conditioning in income's past helps predicting consumption today beyond what past consumption would have allowed.

Given the cointegration vector found in the empirical analysis, we implemented the multivariate structural time-series model in the form suggested by Harvey (1985b) and Koopman et al. (2009). Figure 1 shows the result of this exercise. The consumption series and the trend are very close throughout the whole period, reflecting the fact that agents do update their beliefs about future income, and that the permanent-income theory is probably a reasonable approximation to consumption behavior; see Cochrane (1994) *inter alia*. Also, the cyclical component of consumption varies much more in the pre-WWII era than afterwards.

Next, we present the results of the structural time-series model discussed in Section 3.2, where trends and cycles are estimated imposing that their shocks are independent. Table 2 displays the description of the data in terms of the parameters estimates associated with the log of consumption (2.7) under alternative periods in it. Estimates are obtained for four distinct periods: pre-WWII data – 1901-1941, post-WWII data – 1947-2000, 20th Century data – 1901-2000, and the whole period 1901-2010. It is obvious that uncertainty in the pre-WWII period is much larger than in the post-WWII period. In the pre-WWII era, the variance of the permanent component is about three times that of the post-WWII era. Results for the transitory component are even more striking: about four times.

The estimates of the *total* welfare costs are presented in Table 3. First, there are major differences in results for the pre-WWII and the post-WWII era. This is true regarding the welfare cost of business cycles (associated with transitory shocks), the welfare cost of economic growth (associated with permanent shocks), and the welfare costs of macroeconomic uncertainty (associated with both shocks). These

differences can reach up to 15 times for reasonable parameter values  $-\beta = 0.985$ , and  $\phi = 5$ , for example. Second, regarding the welfare costs of business cycles in the post-WWII period, our results are very similar to those of Lucas, although the methods of estimation are completely different. Third, the welfare costs of economic growth can be twice or three times those of business cycles, while welfare costs of macroeconomic uncertainty can be about 50% larger than those of economic growth.

We now turn our attention to the analysis of the pre-WWII period (1901-1941). For reasonable preference parameter and discount values ( $\beta = 0.985, \phi = 5$ ), welfare costs are 0.31% of consumption if we consider only permanent shocks and 0.58% of consumption if we consider only transitory shocks, which roughly translates into US\$ 60.00 a year and US\$ 120.00 a year, respectively, in current value. In comparison, the post-WWII era is much quieter: welfare costs of economic growth are 0.106% and welfare costs of business cycles are 0.037% – the latter being very close to the estimate in Lucas (0.040%). Results for the whole period 1901-2010 are a combination of those of pre- and post-WWII eras. For reasonable preference parameter and discount values ( $\beta = 0.985, \phi = 5$ ) we get a compensation of 0.48% and 0.27% of consumption, respectively.

We now compare our empirical results with those in Reis (2009). He does not separate the effects of transitory and permanent shocks, i.e., he computes the welfare cost of *all* macroeconomic uncertainty. We compare Table 4 in Reis (sample 1947-2003), where a unit root is imposed for consumption, with our results for  $\lambda_D$  for post-WWII data (sample 1947-2000). Using an ARMA model for the instantaneous growth rate of consumption, Reis finds welfare costs to be roughly between 0.5% and 5% of consumption, whereas we find much lower estimates – between 0.05% and 0.15%. When Reis compared his results to those in Obstfeld (1994), there is also a large difference in estimates, which he attributed to the use of the calibrated effective discount rate  $\rho = \beta^* + (\phi - 1) \ln (1 + \alpha_1)$ , instead of the subjective discount rate  $\beta^*$ , where  $\beta = \exp(-\beta^*)$ . Since  $\rho$  and  $\beta^*$  are identical for  $\phi = 1$ , results in this case are directly comparable: when  $\rho = \beta^* = 0.03$ , and thus  $\beta = 0.97$ , Reis reports a welfare cost of 0.31% of consumption, whereas we find 0.083%, roughly 1/4 of his estimate; for  $\rho = \beta^* = 0.015$ , and thus  $\beta = 0.985$ , we find 0.16%, whereas Reis finds 1.25% for  $\rho = \beta^* = 0.01$ , and 0.61% for  $\rho = \beta^* = 0.02$ , both much higher than our estimate. Thus, there must be an additional source of differences at work here<sup>10</sup>.

Table 4 presents estimates of marginal welfare costs. They are roughly twice the size of welfare costs reported in Table 3. For the pre-WWII era, and reasonable preference parameter and discount values ( $\beta = 0.985, \phi = 5$ ), the marginal welfare costs of economic growth and of business cycles are respectively 0.627% and 1.169% of per-capita consumption. The same figures for the post-WWII era are, respectively, 0.212% and 0.074% of per-capita consumption. The latter can be compared to marginal costs found by Alvarez and Jermann (2004) for 1954-2001: between 0.08% and 0.49% of consumption, when computed at business-cycle frequencies alone. As we argued above, if one does not disentangle the effects of permanent and transitory shocks to consumption, there is the risk of upward biasing the estimate of the welfare costs of business cycles alone. Notwithstanding the slight difference in sample periods in both cases, the estimates in Alvarez and Jermann are higher than our estimate for the welfare costs of business cycles -0.074%.

Results for the whole period 1901-2000 are indeed a combination of those of pre- and post-WWII eras. For reasonable preference parameter and discount values  $(\beta = 0.985, \phi = 5)$  we get a compensation of 0.972% if we consider only permanent shocks. If we take into account only transitory shocks we get 0.54% of per-capita consumption. Extending the sample period up to 2010, which includes the last global recession, makes little difference in welfare-cost estimates.

Testing whether welfare costs are statistically significant can be done for all subsamples employed here. With the exception of welfare costs of economic-growth variation for the 1947-2000 period, all other welfare costs are significantly different

 $<sup>^{10}</sup>$ One possible source is the fact that Reis fits an ARMA model which uses as information set only lagged consumption growth. We use a bi-variate model comprised of consumption and income. Given the evidence of Granger (1969) causality from income to consumption (a t-statistic of -3.67 for income growth in consumption's equation) with post-WWII data, this reduces the variance of shocks to the latter.

from zero.

From the discussion above we can conclude the following. First, current marginal and total welfare cost of business cycles are small – 1947-2010. Hence, it makes little sense to deepen current counter-cyclical policies. Second, from the point of view of a pre-WWII consumer, the marginal welfare costs of business cycles were fairly large. Indeed, for reasonable parameter values ( $\beta = 0.985, \phi = 5$ ) they were 1.169% of consumption in all dates and states of nature. Therefore, from her (his) point of view, it made sense to have had counter-cyclical policies implemented in the post-WWII era.

Last, but not least, a comparison between the welfare costs of business cycles in the pre-WWII and post-WWII periods can give some idea of the effectiveness of counter-cyclical policies which were implemented in the latter period. Considering reasonable parameter values such as  $\beta = 0.985$  and  $\phi = 5$ , the welfare cost of business cycles ( $\lambda_P$ ) decreased from 0.583% to 0.037% of consumption – roughly a factor of 15. The reduction in the marginal welfare costs of business cycles ( $\lambda'_P(0)$ ) are even more impressive: from 1.169% to 0.074% of per-capita consumption. Indeed, if we could credit these reductions in welfare costs to post-WWII counter-cyclical policies – which, by the way, is a big if – it is hard to find any type of implemented economic policy in the name of which it could be claimed such an impressive impact on welfare.

#### 5. Conclusion

Using only standard assumptions on preferences and an econometric approach for modelling consumption, we separate the effects of uncertainty stemming from businesscycle fluctuations and economic growth variation. We model the trend in consumption as a martingale process, while fluctuations about the trend are a stationary and ergodic zero-mean process. Trend and cyclical innovations are assumed to be independent sources of uncertainty. This hypothesis allows the measurement of welfare costs of business cycles and also of economic growth variation.

The whole of the literature chose to work primarily with post-WWII data. How-

ever, for this period, actual consumption is already a result of counter-cyclical policies, and is potentially smoother than what it otherwise would have been in their absence. Because of this, we use four distinct sample periods: pre-WWII data – 1901-1941, post-WWII data – 1947-2000, 20th Century data – 1901-2000, and the whole sample – 1901-2010.

For the estimates of the total welfare costs  $(\lambda_P, \lambda_T, \lambda_D)$ , there are major differences in results for the pre-WWII and the post-WWII era. This is true regarding the welfare cost of business cycles (associated with transitory shocks), the welfare cost of economic growth (associated with permanent shocks), and the welfare costs of macroeconomic uncertainty (associated with both shocks). These differences can reach up to 15 times for reasonable parameter values  $-\beta = 0.985$ , and  $\phi = 5$ , for example. In pre-WWII period (1901-1941), for reasonable preference parameter and discount values ( $\beta = 0.985, \phi = 5$ ), we get welfare costs of 0.310% of consumption if we consider only permanent shocks and 0.608% of consumption if we consider only transitory shocks, which roughly translates into US\$ 60.00 a year and US\$ 120.00 a year, respectively, in current value. In comparison, the post-WWII era is much quieter: welfare costs of economic growth are 0.106% (not significant) and welfare costs of business cycles are 0.037% – the latter being very close to the estimate in Lucas (0.040%).

The estimates of marginal welfare costs  $(\lambda'_P(0), \lambda'_T(0), \lambda'_D(0))$  are roughly twice the size of the total welfare costs. For the pre-WWII era, and reasonable preference parameter and discount values ( $\beta = 0.985, \phi = 5$ ), the marginal welfare costs of economic growth and of business cycles are respectively 0.627% and 1.169% of per-capita consumption. The same figures for the post-WWII era are, respectively, 0.212% and 0.074% of per-capita consumption. The latter can be compared to welfare costs estimated by Alvarez and Jermann (2004). For the 1954-2001 period, they find it to be between 0.08% and 0.49% of consumption, when computed at business-cycle frequencies alone. As we argued above, if one does not disentangle the effects of permanent and transitory shocks to consumption, there is the risk of over-estimating the welfare costs of business cycles alone.

We can conclude the following. First, current marginal and total welfare costs of business cycles are small. Hence, it makes little sense to deepen current countercyclical policies. This is true even including in our sample the data for the last global recession. Second, from the point of view of a pre-WWII consumer, marginal and total welfare costs of business cycles were fairly large. Therefore, from her (his) point of view, it made sense to have had counter-cyclical policies implemented then. Last, a comparison between the welfare costs of business cycles in the pre-WWII and post-WWII period can give some idea of the effectiveness of counter-cyclical policies implemented in the latter period. Considering reasonable parameter values such as  $\beta = 0.985$  and  $\phi = 5$ , the welfare cost of business cycles ( $\lambda_P$ ) decreased from 0.583% to 0.037% of consumption – roughly a factor of 15. Notice that the reduction in the marginal welfare costs of business cycles  $(\lambda'_{P}(0))$  are even more impressive: from 1.169% to 0.074% of per-capita consumption. Indeed, if we could credit these reductions in welfare costs to post-WWII counter-cyclical policies – which, by the way, is a big if - it is hard to find any type of implemented economic policy in the name of which it could be claimed such an impressive impact on welfare.

#### References

- Alvarez, F. and Jermann, U., 2004, "Using Asset Prices to Measure the Cost of Business Cycles," *Journal of Political Economy*, 112(6), pp. 1223-56.
- [2] Alvarez, F. and Jermann, U., 2005, "Using asset prices to measure the persistence of the marginal utility of wealth," *Econometrica*, Vol. 73(6), pp. 1977– 2016.
- [3] Amemiya, Takeshi, 1985, "Advanced Econometrics," Cambridge: Harvard University Press.

- [4] Atkeson, A. and Phelan, C., 1995, "Reconsidering the Cost of Business Cycles with Incomplete Markets", NBER Macroeconomics Annual, 187-207, with discussions.
- [5] Barillas, Francisco, Lars Peter Hansen, and Thomas J. Sargent, 2009, "Doubts or Variability?" forthcoming in the *Journal of Economic Theory*.
- [6] Barro, Robert J., 2009, "Rare Disasters, Asset Prices, and Welfare Costs," American Economic Review, Volume 99, Number 1, pp. 243-264.
- [7] Beveridge, S. and Nelson, C.R., 1981, "A New Approach to Decomposition of Economic Time Series into a Permanent and Transitory Components with Particular Attention to Measurement of the 'Business Cycle'," *Journal of Monetary Economics*, 7, 151-174.
- [8] Campbell, J. 1987, "Does Saving Anticipate Declining Labor Income? An Alternative Test of the Permanent Income Hypothesis," *Econometrica*, vol. 55(6), pp. 1249-73.
- [9] Campbell, John Y. and Deaton, Angus 1989, "Why is Consumption So Smooth?" The Review of Economic Studies 56:357-374.
- [10] Campbell, J.Y. and N.G. Mankiw, 1989, "Consumption, income, and interest rates: reinterpreting the time series evidence," *NBER Macroeconomics Annual*, 4, pp. 185-216.
- [11] Carvalho, Vasco, Andrew Harvey, and Thomas Trimbur, 2007, "A Note on Common Cycles, Common Trends, and Convergence," *Journal of Business and Economic Statistics*, vol. 25(1), pp. 12-20.
- [12] Cochrane, J.H., 1994, "Permanent and Transitory Components of GNP and Stock Prices," *Quarterly Journal of Economics*, 30, 241-265.
- [13] De Santis, Massimiliano, 2009, "Individual Consumption Risk and the Welfare Cost of Business Cycles," American Economic Review, 97(4), pp. 1488-1506.

- [14] Dolmas, J., 1998, "Risk Preferences and the Welfare Cost of Business Cycles", *Review of Economic Dynamics*, 1, 646-676.
- [15] Duarte, A.M., Issler, J.V., and Salvato, M., 2004, "Are Business Cycles all Alike in Europe?," Mimeo., 2005, downloadable from http://www.anpec.org.br/encontro2005/artigos/A05A031.pdf
- [16] Engle, R.F. and Granger C.W.J., 1987, "Cointegration and Error Correction: Representation, Estimation and Testing", *Econometrica*, 55, 251-276.
- [17] Engle, R.F., Hendry, D.F., and Richard, J.F., 1983, "Exogeneity", *Econometrica*, vol. 55, pp. 277-304.
- [18] Galí, J. (1999), "Technology, Employment and the Business Cycle: Do Technology Shocks Explain Aggregate Fluctuations?" American Economic Review, vol. 89(1), pp. 249-271.
- [19] Granger, C.W.J., 1969, "Investigating causal relations by econometric models and cross-spectral methods." *Econometrica*, 37 (3), pp. 424–438.
- [20] Hall, R.E., 1978, "Stochastic Implications of the Life Cycle-Permanent Income Hypothesis: Theory and Evidence," *Journal of Political Economy*, 86, 971-987.
- [21] Hansen, Lars Peter, 2009, "Modeling the Long Run: Valuation in Dynamic Stochastic Economies," NBER Working Paper.
- [22] Hansen, L.P., 2011, "Dynamic Valuation Decomposition Within Stochastic Economies," forthcoming in *Journal of Econometrics*.
- [23] Hansen, Lars Peter, Thomas Sargent and Thomas Tallarini, 1999, "Robust permanent income and pricing," *Review of Economic Studies*, vol. 66, pp. 873– 907.
- [24] Hansen, Lars Peter and José A. Scheinkman (2009), "Long-Term Risk: An Operator Approach," *Econometrica*, 77(1), pp. 177 - 234.

- [25] Harvey, Andrew C., 1985a, "The Econometric Analysis of Time Series," Camelot Press Limited, Southampton.
- [26] Harvey, Andrew C., 1985b, "Trends and Cycles in Macroeconomic Time Series," Journal of Business and Economic Statistics, vol.3, n.3., july 1985.
- [27] Harrison, P. J.; Akran, M. Generalized Exponentially Weighted Regression and Parsimonious Dynamic Linear Modelling. In: *Time Series Analysis: Theory* and Practice (Vol. 3), North-Holland, Amsterdam, 1983.
- [28] Imrohoroglu, Ayse, 1989, "Cost of Business Cycles With Indivisibilities and Liquidity Constraints", Journal of Political Economy, 97 (6), 1364-1383.
- [29] Issler, J.V. and Vahid, F., 2001, "Common Cycles and the Importance of Transitory Shocks to Macroeconomic Aggregates," *Journal of Monetary Economics*, 47(3), pp. 449-475.
- [30] Issler, J.V., Mello Franco-Neto, A.A., and Guillén, O.T.C., 2008, "The Welfare Cost of Macroeconomic Uncertainty in the Post-War Period," *Economics Letters*, Vol. 98, pp. 167-175.
- [31] Johansen, S., 1988, "Statistical Analysis of Cointegration Vectors," Journal of Economic Dynamics and Control, 12, pp. 231-254.
- [32] Johansen, S., 1991, "Estimation and Hypothesis Testing of Cointegrated Vectors in Gaussian Vector Autoregressions", *Econometrica*, vol. 59-6, pp. 1551-1580.
- [33] King, R.G., Plosser, C.I. and Rebelo, S., 1988, "Production, Growth and Business Cycles. II. New Directions", *Journal of Monetary Economics*, 21, 309-341.
- [34] King, R.G., Plosser, C.I., Stock, J.H. and Watson, M.W., 1991, "Stochastic Trends and Economic Fluctuations", *American Economics Review*, 81, 819-840.

- [35] Koopman, S.J., Harvey, A.C., Doornik, J.A., and Shepard, N., 2009, "STAMP 8.2: Structural Time Series Analyser, Modeller and Predictor," London: Timberlake Consultants Ltd.
- [36] Kydland, F. and E. Prescott, 1982, "Time to Build and Aggregate Fluctuations," *Econometrica*, Vol. 50(6), pp. 1345-70.
- [37] Lucas, R., 1987, "Models of Business Cycles," Oxford: Blackwell.
- [38] Melino, A. (2010), "Measuring the cost of economic fluctuations with preferences that rationalize the equity premium," Canadian Journal of Economics, Volume 43, Issue 2, pp. 405–422.
- [39] Nelson, C.R. and Plosser, C., 1982, "Trends and Random Walks in Macroeconomics Time Series," *Journal of Monetary Economics*, 10, 1045-1066.
- [40] Obstfeld, M., 1994, "Evaluating Risky Consumption Paths: The Role of Intertemporal Substitutability," *European Economic Review*, 38, 1471-1486.
- [41] Otrok, C., 2001, "On Measuring the Welfare Cost of Business Cycles," Journal of Monetary Economics, 47, 61-92.
- [42] Pemberton, J., 1996, "Growth Trends, Cyclical Fluctuations, and Welfare with Non-Expected Utility Preferences," *Economic Letters*, 50, 387-392.
- [43] Phelps, E.S. (1967), "Phillips Curves, Expectations of Inflation and Optimal Unemployment over Time," *Economica*, 34.
- [44] Phelps, E.S. (1968), "Money Wage Dynamics and Labor Market Equilibrium," Journal of Political Economy, 76.
- [45] Phelps, E.S. et al. (1970), "Microeconomic Foundations of Employment and Inflation," New York: Norton.
- [46] Phillips, P.C.B. and V. Solo, 1992, "Asymptotics for linear processes," Annals of Statistics, vol. 20, pp. 971-1001.

- [47] Proietti, T., 1997, "Short-run Dynamics in Cointegrated Systems", Oxford Bulletin of Economics and Statistics, 59 (3), 405-422.
- [48] Reis, R., 2009, "The time-series properties of aggregate consumption: implications for the costs of fluctuations," *Journal of the European Economic Association*, Vol. 7(4), pp. 722–753.
- [49] Tallarini Jr., T.D., 2000, "Risk-sensitive Real Business Cycles", Journal of Monetary Economics, 45, 507-532.
- [50] Stock, J. and Watson, M., 1988, "Testing for Common Trends," Journal of the American Statistical Association, 83, pp. 1097-1109.
- [51] Vahid, F. and Engle, R.F., 1993, "Common Trends and common Cycles", Journal of Applied Econometrics, 8, 341-360.
- [52] Van Nieuwerburgh, Stijn, Lustig, Hanno N. and Verdelhan, Adrien, 2010, "The Wealth-Consumption Ratio." NYU Working Paper No. FIN-08-045.
- [53] Van Wincoop, E., 1994, "Welfare Gains From International Risksharing", Journal of Monetary Economics, 34, 175-200.
- [54] Zellner, A. (1992), Statistics, Science and Public Policy, Journal of the American Statistical Association, 87, 1–6.

Table 1:	Cointegration te	est – Johai	nsen (1988, 1991)	1) Technie	que
Cointegrating		Trace	5~%	$\lambda_{\max}$	5~%
Vectors under ${\cal H}_0$	Eigenvalues	Stat.	Crit. Value	Stat.	Crit. Value
None	0.150	16.44	15.41	16.26	14.07
At most 1	0.0018	0.18	3.76	0.18	3.76
Estimate of the coint $\epsilon$	egrating vector i	$s \cdot (-1 \ 1 \ 0)$	)05)		

 $H_0: \beta' = (-1, 1)$ , conditional on r = 1, p-value = 0.831.



Figure 1: Real Consumption and Income per-capita



Figure 2: Consumption and Consumption Trends Computed in Different Sub-samples (in logs)



Figure 3: Consumption Cycles Computed in Different Sub-samples (in logs)

Table 2: Co	Table 2: Consumption – Parameter Estimates in Equations (2.7) and (2.8)								
	1901-2000	1901-1941	1947-2000	1901-2010					
$\ln \widehat{(1+\alpha_1)}$	0.0195	0.0152	0.0217	0.0188					
	(0.0013)	(0.0027)	(0.0009)	(0.0011)					
$\widehat{\sigma_{11}}$	0.0001843	9.71885E-05	4.51548E-05	0.000140286					
	(0.0000854)	(4.06191E-05)	(3.88781E-05)	(0.0000663)					
$\widehat{\widetilde{\sigma}_{22}}$	0.0010802	0.0023237	0.0001482	0.0011765					
	(0.0004640)	(0.0010908)	(0.0000257)	(0.0004671)					

Note: Standard errors in parenthesis.

		$\phi = 20$		0.2906	(0.1433)	0.3113	(0.1538)	0.3262	(0.1614)		1.4775	(0.4956)	1.4984	(0.4990)	1.5135	(0.5015)		1.1834	(0.4726)	1.1834	(0.4726)	1.1834	(0.4726)	
	2010	$\phi = 10$		0.2898	(0.1394)	0.3260	(0.1571)	0.3545	(0.1711)		0.8815	(0.2742)	0.9179	(0.2837)	0.9466	(0.2919)		0.5900	(0.2349)	0.5900	(0.2349)	0.5900	(0.2349)	
	1901-3	$\phi = 5$		0.2623	(0.1248)	0.3212	(0.1531)	0.3754	(0.1792)		0.5576	(0.1717)	0.6167	(0.1933)	0.6711	(0.2147)		0.2946	(0.1171)	0.2946	(0.1171)	0.2946	(0.1171)	
of $(\beta, \phi)$		$\phi = 1$		0.1334	(0.0631)	0.2351	(0.1112)	0.4617	(0.2187)		0.1923	(0.0673)	0.2941	(0.1137)	0.5208	(0.2201)		0.0588	(0.0234)	0.0588	(0.0234)	0.0588	(0.0234)	
erent Values		$\phi = 20$		0.3699	0.1811	0.3957	0.1942)	0.4142	0.2037)		1.4600	0.5051)	1.4861	0.5101)	1.5048	0.5140)		1.0861	0.4690)	1.0861	0.4690)	1.0861	0.4690)	
%) for diffe	000	$\phi = 10$		0.3695	(0.1752) (	0.4144	(0.1969) (	0.4496	(0.2140) (		0.9131	(0.2930) (	0.9582	(0.3067) (	0.9936	(0.3181) (		0.5416	(0.2332) (	0.5416	(0.2332) (	0.5416	(0.2332) (	1
certainty (	1901-2	$\phi = 5$		0.3364	(0.1574) (	0.4101	(0.1922) (	0.4773	(0.2241) (		0.6078	(0.1963) (	0.6816	(0.2254) (	0.749	(0.2533) (		0.2704	(0.1163) (	0.2704	(0.1163) (	0.2704	(0.1163) (	
economic Un		$\phi = 1$		0.1752	(0.0813)	0.3090	(0.1435)	0.6069	(0.2822)		0.2293	(0.0846)	0.3632	(0.1454)	0.6613	(0.2833)		0.0540	(0.0232)	0.0540	(0.0232)	0.0540	(0.0232)	
ch, and Macro		$\phi = 20$		0.0776	(0.0670)	0.0824	(0.0712)	0.0859	(0.0742)		0.2260	(0.0718)	0.2308	(0.0757)	0.2343	(0.0786)		0.1483	(0.0255)	0.1483	(0.0255)	0.1483	(0.0255)	
omic Growt	2000	$\phi = 10$		0.0812	(0.0696)	0.0902	(0.0773)	0.0971	(0.0833)		0.1554	(0.0708)	0.1644	(0.0784)	0.1712	(0.0843)		0.0741	(0.0128)	0.0741	(0.0128)	0.0741	(0.0128)	
/cles, Econe	1947-2	$\phi = 5$		0.0764	(0.0653)	0.0919	(0.0786)	0.1056	(0.0903)		0.1135	(0.0657)	0.1290	(0.0789)	0.1427	(0.0906)		0.0371	(0.0064)	0.0371	(0.0064)	0.0371	(0.0064)	
f Business Cy		$\phi = 1$		0.0429	(0.0366)	0.0756	(0.0645)	0.1484	(0.1267)		0.0503	(0.0366)	0.0830	(0.0645)	0.1558	(0.1267)		0.0074	(0.0013)	0.0074	(0.0013)	0.0074	(0.0013)	
elfare Costs o		$\phi = 20$		0.2477	(0.1071)	0.2683	(0.1163)	0.2835	(0.1230)		2.6044	(1.1246)	2.6255	(1.1258)	2.6410	(1.1267)		2.3509	(1.1165)	2.3509	(1.1165)	2.3509	(1.1165)	
Table 3: W	1941	$\phi = 10$		0.2378	(0.1008)	0.2725	(0.1157)	0.3007	(0.1279)		1.4092	(0.5624)	1.4443	(0.5655)	1.4729	(0.5684)		1.1686	(0.5518)	1.1686	(0.5518)	1.1686	(0.5518)	
	1901-	$\phi = 5$		0.2058	(0.0865)	0.2587	(0.1089)	0.3101	(0.1307)		0.7896	(0.2883)	0.8429	(0.2960)	0.8945	(0.3049)		0.5826	(0.2743)	0.5826	(0.2743)	0.5826	(0.2743)	
		$\phi = 1$		0.0924	(0.0386)	0.1628	(0.0681)	0.3196	(0.1338)		0.2087	(0.0670)	0.2793	(0.0874)	0.4362	(0.1447)		0.1163	(0.0546)	0.1163	(0.0546)	0.1163	(0.0546)	
			$\lambda_T(\cdot)$	eta=0.950		eta=0.971		$\beta = 0.985$		$\lambda_D(\cdot)$	$\beta = 0.950$		eta=0.971		$\beta = 0.985$		$\lambda_P(\cdot)$	eta=0.950		eta=0.971		eta=0.985		

17)
2
Ţ
an
),
14
(2.
÷.
12
(2
$\mathbf{US}$
101
ъ
eqn
ц
y, i
el.
tiv
) ec
est
Ē,
en
.2.
e
aı
$\lambda_D$
р
and
T, and
, $\lambda_T$ , and
$\lambda_P, \lambda_T, \text{ and }$
or $\lambda_P$ , $\lambda_T$ , and
s for $\lambda_P$ , $\lambda_T$ , and
las for $\lambda_P$ , $\lambda_T$ , and
nulas for $\lambda_P$ , $\lambda_T$ , and
ormulas for $\lambda_P$ , $\lambda_T$ , and
Formulas for $\lambda_P$ , $\lambda_T$ , and
0). Formulas for $\lambda_P$ , $\lambda_T$ , and
100). Formulas for $\lambda_P$ , $\lambda_T$ , and
by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
d by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
died by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
tiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
nultiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
(multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
sis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
hesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
m thesis (multiplied by 100). Formulas for $\lambda_{P}, \lambda_{T},$ and
arenthesis (multiplied by 100). Formulas for $\lambda_{P}$ , $\lambda_{T}$ , and
parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
we in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
s are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
ors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
d Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
lard Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
indard Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
Standard Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
:: Standard Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and
ote: Standard Errors are in parenthesis (multiplied by 100). Formulas for $\lambda_P$ , $\lambda_T$ , and

		Table 1001	e 4: Margi:	nal Welfare C	osts of Busine	ss Cycles,	Economic	Growth, and	Macroeconom	nic Uncerta	inty (%) ft	or different Val	ues of $(\beta, \phi)$	1001	010	
		1901-	-1941			1947-	2000			1901-	2000			-1001	2010	
	$\phi = 1$	$\phi = 5$	$\phi = 10$	$\phi = 20$	$\phi = 1$	$\phi = 5$	$\phi = 10$	$\phi = 20$	$\phi = 1$	$\phi = 5$	$\phi = 10$	$\phi = 20$	$\phi = 1$	$\phi = 5$	$\phi = 10$	$\phi = 20$
$\lambda_T'(\cdot)$																
eta=0.950	0.1850	0.4148	0.4837	0.5144	0.0859	0.1533	0.1634	0.1572	0.3514	0.6818	0.7595	0.786	0.2673	0.5300	0.5922	0.6092
	(0.0775)	(0.1757)	(0.2085)	(0.2308)	(0.0733)	(0.1315)	(0.1409)	(0.1375)	(0.1635)	(0.3233)	(0.3701)	(0.4079)	(0.1266)	(0.2549)	(0.2909)	(0.3144)
eta = 0.971	0.3265	0.5226	0.5555	0.5585	0.1514	0.1844	0.1815	0.1671	0.6209	0.8333	0.8542	0.8432	0.4720	0.6503	0.6676	0.6540
	(0.1369)	(0.2221)	(0.2404)	(0.2517)	(0.1294)	(0.1583)	(0.1567)	(0.1463)	(0.2897)	(0.3968)	(0.4182)	(0.4400)	(0.2241)	(0.3139)	(0.3293)	(0.3390)
$\beta = 0.985$	0.6423	0.6274	0.6142	0.5911	0.2974	0.2121	0.1955	0.1742	1.2251	0.9721	0.9287	0.8846	0.9298	0.7616	0.7271	0.6866
	(0.2702)	(0.2675)	(0.2666)	(0.2672)	(0.2544)	(0.1821)	(0.1689)	(0.1526)	(0.5749)	(0.4648)	(0.4565)	(0.4634)	(0.4435)	(0.3687)	(0.3598)	(0.3570)
$\lambda'_D(\cdot)$																
$\beta = 0.950$	0.4181	1.5883	2.8460	5.296	0.1007	0.2276	0.3119	0.4545	0.4599	1.2271	1.8539	2.9871	0.3853	1.1231	1.7826	3.0046
	(0.1343)	(0.5819)	(1.1420)	(2.3099)	(0.0734)	(0.1322)	(0.1435)	(0.1471)	(0.1702)	(0.4010)	(0.6027)	(1.0426)	(0.1352)	(0.3486)	(0.5591)	(1.0146)
eta=0.971	0.5599	1.6973	2.9195	5.3422	0.1663	0.2587	0.3301	0.4644	0.7297	1.3793	1.9496	3.0456	0.5902	1.2442	1.8589	3.0505
	(0.1757)	(0.5985)	(1.1493)	(2.3132)	(0.1294)	(0.1589)	(0.1590)	(0.1554)	(0.2937)	(0.4631)	(0.6344)	(1.0566)	(0.2292)	(0.3944)	(0.5808)	(1.0233)
eta=0.985	0.8765	1.8034	2.9795	5.3763	0.3123	0.2863	0.3441	0.4715	1.3345	1.5189	2.0248	3.0879	1.0487	1.3560	1.9192	3.0838
	(0.2923)	(0.6177)	(1.1560)	(2.3159)	(0.2545)	(0.1827)	(0.1711)	(0.1614)	(0.5774)	(0.5233)	(0.6610)	(1.0673)	(0.4465)	(0.4400)	(0.5993)	(1.0300)
$\lambda_P'(\cdot)$																
$\beta = 0.950$	0.2326	1.1686	2.3509	4.7571	0.0148	0.0741	0.1483	0.2968	0.1081	0.5416	1.0861	2.1839	0.1177	0.5900	1.1834	2.3809
	(0.1343)	(0.5518)	(1.1165)	(2.2854)	(0.0734)	(0.0128)	(0.0255)	(0.0511)	(0.1702)	(0.2332)	(0.4690)	(0.9482)	(0.1352)	(0.2349)	(0.4726)	(0.9564)
eta=0.971	0.2326	1.1686	2.3509	4.7571	0.0148	0.0741	0.1483	0.2968	0.1081	0.5416	1.0861	2.1839	0.1177	0.5900	1.1834	2.3809
	(0.1343)	(0.5518)	(1.1165)	(2.2854)	(0.0734)	(0.0128)	(0.0255)	(0.0511)	(0.1702)	(0.2332)	(0.4690)	(0.9482)	(0.1352)	(0.2349)	(0.4726)	(0.9564)
eta=0.985	0.2326	1.1686	2.3509	4.7571	0.0148	0.0741	0.1483	0.2968	0.1081	0.5416	1.0861	2.1839	0.1177	0.5900	1.1834	2.3809
	(0.1343)	(0.5518)	(1.1165)	(2.2854)	(0.0734)	(0.0128)	(0.0255)	(0.0511)	(0.1702)	(0.2332)	(0.4690)	(0.9482)	(0.1352)	(0.2349)	(0.4726)	(0.9564)
Note: Stand	ard Errors a	re in parenth	esis (multipl	ied by 100). For	mulas for $\lambda'_P$ , $\lambda$	$I'_T$ , and $\lambda'_D$ a	re given, resl	pectively, in eq	lations (2.18), (2	2.19), and (2	20).					

3
and
5.19
10
.18
3
equations
E.
, respectively
are given.
$\sim^{\Omega}$
and .
$\lambda_{T}'$
,á
.5
Formulas f
00).
y 1
(multiplied 1
parenthesis
rs are in
tandard Erro
Note: S

## **Banco Central do Brasil**

### Trabalhos para Discussão

Os Trabalhos para Discussão do Banco Central do Brasil estão disponíveis para download no website http://www.bcb.gov.br/?TRABDISCLISTA

#### **Working Paper Series**

The Working Paper Series of the Central Bank of Brazil are available for download at http://www.bcb.gov.br/?WORKINGPAPERS

245	Pesquisa Trimestral de Condições de Crédito no Brasil Clodoaldo Aparecido Annibal e Sérgio Mikio Koyama	Jun/2011
246	Impacto do Sistema Cooperativo de Crédito na Eficiência do Sistema Financeiro Nacional Michel Alexandre da Silva	Ago/2011
247	Forecasting the Yield Curve for the Euro Region Benjamim M. Tabak, Daniel O. Cajueiro and Alexandre B. Sollaci	Aug/2011
248	Financial regulation and transparency of information: first steps on new land Helder Ferreira de Mendonça, Délio José Cordeiro Galvão and Renato Falci Villela Loures	Aug/2011
249	<b>Directed clustering coefficient as a measure of systemic risk in complex banking networks</b> <i>B. M. Tabak, M. Takami, J. M. C. Rocha and D. O. Cajueiro</i>	Aug/2011
250	<b>Recolhimentos Compulsórios e o Crédito Bancário Brasileiro</b> Paulo Evandro Dawid e Tony Takeda	Ago/2011
251	Um Exame sobre como os Bancos Ajustam seu Índice de Basileia no Brasil Leonardo S. Alencar	Ago/2011
252	<b>Comparação da Eficiência de Custo para BRICs e América Latina</b> Lycia M. G. Araujo, Guilherme M. R. Gomes, Solange M. Guerra e Benjamin M. Tabak	Set/2011
253	<b>Bank Efficiency and Default in Brazil: causality tests</b> Benjamin M. Tabak, Giovana L. Craveiro and Daniel O. Cajueiro	Oct/2011
254	<b>Macroprudential Regulation and the Monetary Transmission</b> <b>Mechanism</b> <i>Pierre-Richard Agénor and Luiz A. Pereira da Silva</i>	Nov/2011
255	<b>An Empirical Analysis of the External Finance Premium of Public</b> <b>Non-Financial Corporations in Brazil</b> <i>Fernando N. de Oliveira and Alberto Ronchi Neto</i>	Nov/2011
256	<b>The Self-insurance Role of International Reserves and</b> <b>the 2008-2010 Crisis</b> <i>Antonio Francisco A. Silva Jr</i>	Nov/2011

257	Cooperativas de Crédito: taxas de juros praticadas e fatores de viabilidade Clodoaldo Aparecido Annibal e Sérgio Mikio Koyama	Nov/2011
258	<b>Bancos Oficiais e Crédito Direcionado – O que diferencia o mercado de crédito brasileiro?</b> <i>Eduardo Luis Lundberg</i>	Nov/2011
259	The impact of monetary policy on the exchange rate: puzzling evidence from three emerging economies <i>Emanuel Kohlscheen</i>	Nov/2011
260	<b>Credit Default and Business Cycles: an empirical investigation of Brazilian retail loans</b> Arnildo da Silva Correa, Jaqueline Terra Moura Marins, Myrian Beatriz Eiras das Neves and Antonio Carlos Magalhães da Silva	Nov/2011
261	<b>The relationship between banking market competition and risk-taking:</b> <b>do size and capitalization matter?</b> <i>Benjamin M. Tabak, Dimas M. Fazio and Daniel O. Cajueiro</i>	Nov/2011
262	<b>The Accuracy of Perturbation Methods to Solve Small Open</b> <b>Economy Models</b> <i>Angelo M. Fasolo</i>	Nov/2011
263	<b>The Adverse Selection Cost Component of the Spread of</b> <b>Brazilian Stocks</b> <i>Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo and José</i> <i>Valentim Machado Vicente</i>	Dec/2011
264	Uma Breve Análise de Medidas Alternativas à Mediana na Pesquisa de Expectativas de Inflação do Banco Central do Brasil Fabia A. de Carvalho	Jan/2012
265	<b>O Impacto da Comunicação do Banco Central do Brasil sobre o Mercado Financeiro</b> <i>Marcio Janot e Daniel El-Jaick de Souza Mota</i>	Jan/2012
266	<b>Are Core Inflation Directional Forecasts Informative?</b> <i>Tito Nícias Teixeira da Silva Filho</i>	Jan/2012
267	Sudden Floods, Macroprudention Regulation and Stability in an Open Economy PR. Agénor, K. Alper and L. Pereira da Silva	Feb/2012
268	<b>Optimal Capital Flow Taxes in Latin America</b> João Barata Ribeiro Blanco Barroso	Mar/2012
269	<b>Estimating Relative Risk Aversion, Risk-Neutral and Real-World</b> <b>Densities using Brazilian Real Currency Options</b> <i>José Renato Haas Ornelas, José Santiago Fajardo Barbachan and Aquiles</i> <i>Rocha de Farias</i>	Mar/2012
270	<b>Pricing-to-market by Brazilian Exporters: a panel cointegration approach</b> <i>João Barata Ribeiro Blanco Barroso</i>	Mar/2012
271	<b>Optimal Policy When the Inflation Target is not Optimal</b> Sergio A. Lago Alves	Mar/2012

272	<b>Determinantes da Estrutura de Capital das Empresas Brasileiras: uma abordagem em regressão quantílica</b> Guilherme Resende Oliveira, Benjamin Miranda Tabak, José Guilherme de Lara Resende e Daniel Oliveira Cajueiro	Mar/2012
273	<b>Order Flow and the Real: Indirect Evidence of the Effectiveness of</b> <b>Sterilized Interventions</b> <i>Emanuel Kohlscheen</i>	Apr/2012
274	<b>Monetary Policy, Asset Prices and Adaptive Learning</b> <i>Vicente da Gama Machado</i>	Apr/2012
275	A geographically weighted approach in measuring efficiency in panel data: the case of US saving banks Benjamin M. Tabak, Rogério B. Miranda and Dimas M. Fazio	Apr/2012
276	A Sticky-Dispersed Information Phillips Curve: a model with partial and delayed information Marta Areosa, Waldyr Areosa and Vinicius Carrasco	Apr/2012
277	<b>Trend Inflation and the Unemployment Volatility Puzzle</b> Sergio A. Lago Alves	May/2012
278	Liquidez do Sistema e Administração das Operações de Mercado Aberto Antonio Francisco de A. da Silva Jr.	Maio/2012
279	<b>Going Deeper Into the Link Between the Labour Market and Inflation</b> <i>Tito Nícias Teixeira da Silva Filho</i>	May/2012
280	Educação Financeira para um Brasil Sustentável Evidências da necessidade de atuação do Banco Central do Brasil em educação financeira para o cumprimento de sua missão Fabio de Almeida Lopes Araújo e Marcos Aguerri Pimenta de Souza	Jun/2012
281	<b>A Note on Particle Filters Applied to DSGE Models</b> Angelo Marsiglia Fasolo	Jun/2012
282	<b>The Signaling Effect of Exchange Rates: pass-through under dispersed information</b> <i>Waldyr Areosa and Marta Areosa</i>	Jun/2012
283	<b>The Impact of Market Power at Bank Level in Risk-taking:</b> <b>the Brazilian case</b> <i>Benjamin Miranda Tabak, Guilherme Maia Rodrigues Gomes and Maurício</i> <i>da Silva Medeiros Júnior</i>	Jun/2012